

Received April 10, 2022, accepted May 21, 2022, date of publication May 30, 2022, date of current version June 7, 2022. *Digital Object Identifier* 10.1109/ACCESS.2022.3179047

Hardware-Assisted Machine Learning in Resource-Constrained IoT Environments for Security: Review and Future Prospective

GEORGIOS KORNAROS^{ID}

Department of Electrical and Computer Engineering, Hellenic Meditteranean University, 71410 Iraklio, Greece

e-mail: kornaros@hmu.gr

This work was supported in part by the European Union (EU) Horizon 2020 Project AVANGARD (advanced manufacturing solutions tightly aligned with business needs) under Agreement 869986.

ABSTRACT As the Internet of Things (IoT) technology advances, billions of multidisciplinary smart devices act in concert, rarely requiring human intervention, posing significant challenges in supporting trusted computing and user privacy, as well as protecting against attacks such as spoofing, denial of service (DoS), jamming, and eavesdropping. To tackle attacks on the IoT and cyber-physical ecosystem, many intrusion detection and security approaches have been presented in the literature. Machine learning (ML) based intrusion and anomaly detection has lately gained traction due to its capacity to cope with encrypted and rapidly developing threat techniques. This work investigates into machine learning (ML) and deep learning (DL) methodologies for IoT device security and examine the benefits, drawbacks, and potential. To protect an IoT infrastructure, various solutions look into hardware-based methods for ML-based IoT authentication, access control, secure offloading, and malware detection schemes. This review aims to illuminate the value of various approaches for addressing IoT security in a truly effective, flexible, and seamless manner, as well as to provide answers to questions about tradeoffs in integrating accelerators and customizing embedded device architectures for effective use of ML-based methods.

INDEX TERMS AI-based IoT security, hardware-based machine learning, IoT intrusion detection, trusted embedded devices.

I. INTRODUCTION

To ensure end-to-end secure cyberphysical systems with Internet-of-Things (IoT) infrastructures, one important parameter ignored till recently, necessitates the integration of a substantial level of cyber protection of IoT end devices, namely, of the microcontroller units (MCUs) and their interconnection networks [1]–[5]. To tackle software and hardware threats, modern embedded Systems-on-Chip (SoCs) are designed considering compliance with the requirements of the ARM Trusted Base System architecture [6], i.e., base SoC isolation, cryptography trusted boot, and debug protection. ARM TrustZone hardware organization is used to isolate the target application from other applications at runtime by providing a partitioning of internal and external memory into trusted and non-trusted worlds. By leveraging this principle, numerous typical security features for IoT connected devices are fulfilled, such as secure boot, secure firmware installation, cryptographic accelerators, secure data storage and secure firmware update [7]. Additional protection of applications and data on such devices includes several techniques ranging from software to specialized circuitry, such as emerging instruction set extensions, ARM's Branch Target Identification (BTI) and Memory Tagging Extension (MTE) to mitigate memory-related security bugs [8], [9]. It is equally of growing concern to ensure built-in active tamper detection, by reducing the vulnerability surface of encryption of symmetric accelerators (e.g., AES) and asymmetric public-key accelerators (PKA) against attacks with side-channel analysis (SCA), secure hardware and independent keys for persistent data storage. Additionally, security countermeasures may also support internal monitoring of perturbation attacks to erase secret data, according to PCI security standards council requirements for end-point applications and data [10]. Attestation along with Hardware Security Managers (HSM) and Physically Unclonable Functions (PUF) have also been

The associate editor coordinating the review of this manuscript and approving it for publication was Sathish Kumar¹⁰.

proposed for integrity protection [11]–[13]. HSM units are employed to orchestrate not only the key distribution, but also cryptography related operations such as authentication, cryptography-based trusted boot and debug protection. Moreover, security strategies extend beyond hardware design protection (e.g., against the insertion of hardware Trojans during the production phase, through netlist obfuscation provided by logic locking) through firmware and operating system layers, such as various secure, trusted, and verified microkernel architectures [14].

Nonetheless, IoT devices are mostly restricted devices with inadequate tamper-resistant and tamper-detection methods, allowing connected devices to leak personal data, for example, by allowing modified firmware to access authentication credentials. With the rapid proliferation of Internet of Things (IoT) devices and cloud systems, most cyber-physical systems' quickly increasing attack surface can scarcely keep attacks from multiplying in quantity and sophistication, as seen in Fig 1. IoT-enabled cyber-physical systems (CPSs) in factories, smart grids, and automobiles now have a variety of communication interfaces, remote monitoring, and software-over-the-air administration capabilities [2], [15]. In essence, Industrial IoT (IIoT) networks have a broad attack surface, making a covert channel more difficult to detect. As a result, most intrusion detection and prevention systems (IDS/IPS) rely on signature originality or signatures that have not been tampered with. Meanwhile, traditional IT network isolation is no longer feasible for IIoT networks. This is due to cautious malware design with obfuscation, or attempts to probe or otherwise manipulate devices and network, can easily bypass signature-based IDS which use matching patterns. Smarter and more advanced boundary control and auditing of access in needed against the trust boundaries of HoT networks [16]. To address such challenges, artificial intelligence (AI) is promising effective solutions pertaining to security.

Essentially, security becomes increasingly complex as the attack surface of computing things increases. Because of the restricted things resources available, key concepts and common security mechanisms may need to be shared throughout layers of security solutions for each one [20]-[23]. Even the integration of micro-architectural features in the latest processors may extend the scope for new side-channel attacks. Performance counters, for example, can indicate branch misses events to aid successful attacks on asymmetric ciphers like RSA, as demonstrated recently [24]. Meanwhile, hostile attackers are becoming more sophisticated, frequent, and automated, even using AI-based approaches to automate IoT security breaches and enable more successful but also less detectable attacks [25], [26]. For instance, recent developments reveal machine learning framework for side-channel attacks on asymmetric cryptography, such as RSA and ECC, that analyzes leakage in multiple side-channel traces, identifying the best trace for key retrieval on a 32-bit ARM Cortex-M4 microcontroller [27].

A. MACHINE LEARNING FOR IOT SECURITY

Fundamentally, using machine learning methodologies involves a threefold scope, (i) to facilitate an effective attack against an IoT infrastructure by exploiting a hardware, software or network vulnerability; (ii) to establish a robust and automated detection and protection system against malware, side-channel threats, fault attacks, and other threats; and (iii) to create the need to implement countermeasures against adversaries to the ML-based techniques themselves.

Machine learning and deep learning algorithms are rapidly being used in cybersecurity applications such as intrusion and virus detection, user authentication (e.g., biometrics), and user privacy. These advanced learning methods may be used to evaluate and learn from underlying IoT data in order to enhance threat assessment and attack detection, and thereby identify breaches in the IoT ecosystem. Deep learning approaches that adapt and evolve at the same time can not avoid sophisticated threats like entity or object profiling, as well as possible interdependent vulnerabilities and exploits. Deep learning can significantly change the cybersecurity landscape. For example, to improve traditional techniques that use pattern-matching to detect malware, such as by using register values and states to identify original identity of industrial embedded devices [28]. These pattern matching solutions can barely match the increasing rate of new attacks and variants. Sophisticated malware has been able to bypass or infiltrate network and end-point detection strategies, thus continously sporting significant cyber-attacks. Additionally, a huge number of IoT devices are lacking in processing power and storage capacity to run security solutions and maintain databases of threat and malware signatures to protect them against threats. On top, even detection methods based on observing anomalies present weaknesses, since activities which users rarely perform may also be classified as an anomaly [29]. In this scope, deep learning can be leveraged to learn and evolve new defense mechanisms using all available data and address the growing cybersecurity challenge [30], [31].

Essentially, pertaining to security of IoT devices and networks, the emerging ML and DNN techniques and branches (e.g., reinforcement learning (RL), Long short term memory (LSTM), generative adversarial network (GAN)) bring the following benefits.

- Machine learning methods help in automated threat detection and prevention by addressing complexity of modeling an indefinite space of malicious behaviours, and by integrating analysis, detection and protection systems.
- Machine learning methods can manage huge number of devices, to navigate their firmware updates and security patches; AI-driven policy and update management can help for firmware updates and patches to apply in all devices in a timely fashion.
- Machine learning for cybersecurity is scalableindependent as it makes it possible for a system to

IEEEAccess



FIGURE 1. Threat model involving a wide spectrum of attacks spreading mainly in IoT authentication, access control, secure offloading, and malware detection, in IoT infrastructures facilitating integration between the physical world and computer communication networks, and applications (apps) [17]–[19].

learn by its own experience as it grows and self-tune to become increasingly efficient and effective.

These advantages, especially, are more valuable in view of combining ML and DNN methods with the increased difficulty to tamper hardware-based techniques. As the security of modern embedded computing devices raises extensive concerns, hardware-based monitors and countermeasures offer increased guarantees when developed and deployed to thwart various cyber attacks. Moreover, hardware-based detection techniques require smaller overhead for resource and latency compared to the software-based counterparts. Several such techniques heavily utilize machine learning (ML) techniques [32]-[34], thus attempting to raise a strong defense umbrella against the numerous threats and attacks. However, the landscape of securing IoT environments using ML methods includes numerous challenges, stemming from the subtle attributes of the various attacks,¹ combined with software and hardware circuitry complexity with security-weak surfaces. This marriage of ML algorithms with secure- and trusted- conscious methods, spanning hardware and software layers, is proving to grow as a two-edged sword in IoT environments [35], as analyzed next in this article.

B. RELATED SURVEYS

Prior works provide surveys that deliver insight into several related topics, without delivering though a unified, comprehensive view on modern research efforts in ML and DNN methods combined with microarchitectural techniques for secure and trusted edge computing. Table 1 summarizes distinctive surveys on secure architectures for trusted computations, on advancements on machine learning practices in IoT and on the intersection between intrusion detection, between hardware acceleration methods of machine learning and on emerging IoT infrastructures.

To the best of my knowledge, this investigation work is a systematic comprehensive review that analyzes different strategies and presents the effectiveness and practical perspectives of machine learning powered methods assisted by hardware techniques and accelerators for the security of IoT devices and systems. In particular, we discuss these techniques, their merits and drawbacks, summarize strengths and weaknesses in hardware-based ML domain for intrusion detection research and suggest future research challenges. The aim of this work is, first, to showcase if the the gap between the capabilities of machine learning (ML) and deep learning (DL) and the requirements of the IoT resourceconstrainted environment can be effectively bridged. This analysis is balanced against today's and emerging cuttingedge microarchitectural advancements, with a view towards addressing the security challenges of the IoT ecosystems. Figure 2 shows the concepts, dimensions investigated in this article, with particular focus on prominent hardware solutions that leverage ML methods towards securing IoT devices.

The rest of the paper is organized as follows. Section II discusses anomaly detection for IoT devices and ecosystem. Section III surveys the literature ML-based hardware methods for security investigating the marriage between ML for systems protection and especially forensics for IoT systems. Section IV reviews and analyzes research and industrial techniques for enhancing IoT security at the edge while bringing ML and DL in support as well. Then, section V provides insight on the effectiveness of the various trends and techniques and identify research gaps that deserve further research efforts. Section VI presents conclusions and suggests future research directions.

¹ for example, beyond common malware, rootkit attacks may opt for code injection, function pointer hooking, direct kernel object manipulation.

TABLE 1. Background research contributions presenting surveys, analyses, taxonomies and future perspectives.

Work	Research Objectives and Dimensions
Chaabouni et al., [1]	IoT security threats and challenges classification for IoT networks, with a focus on network intrusion detection systems (NIDSs)
Luo et al., [17]	State-of-the-art deep learning-based anomaly detection (DLAD) methods for cyber-physical systems
Maene et al., [36]	Trusted computing architectures with guarantees and protection against software-level attackers
Merenda et al., [37]	Review of implementing edge machine learning on Internet of Things devices
Xu et al., [38]	State-of-the-art hardware-based attacks on DNN with focus on hardware Trojans
Mitchel et al., [39]	Classification of intrusion detection system (IDS) techniques in cyber-physical systems (CPS); two design dimensions are
	employed, detection method and audit data (e.g., system calls, traffic from a network interface, hearsay reputation scores)
Al-Garadi et al., [40]	Survey of ML methods and recent advances in DL methods to develop enhanced security methods for individual IoT scenarios
Huang et al., [41]	ML-based approaches against HT attacks hardware Trojan defense (HTD) reference model
Hussain et al., [42]	Role of machine learning in general and deep learning in particular for IoT security
Khan et al., [43]	End-to-end security threats, and models analysis along with lightweight cryptographic protocol analysis
Khraisat et al., [44]	Review of IoT IDS methodology, deployment strategy, validation strategy, dataset and technologies, advantages and limitations
Sidhu et al., [45]	Review and taxonomy of hardware Trojans (HTs) for IoT devices and protection against them
Shawahna et al., [46]	Survey of acceleration of deep learning (DL), convolution neural networks (CNNs) on field programmable gate arrays (FPGAs)
Bochie et al., [47]	Exploration of potential of deep learning techniques in different networks, IoT, sensor, mobile, industrial and vehicular networks
Olowonomi et al., [48]	Challenges in applying AI and ML in cybersecurity of CPSs and methods for defense against adversarial attacks
Rosenberg et al., [49]	Survey and taxonomy of adversarial attacks and defenses against classifiers used in the cyber security domain





II. ANOMALY DETECTION IN IOT EMBEDDED SYSTEMS, CHALLENGES AND OPPORTUNITIES

Anomaly detection methods are developed to mitigate various threats, such as false data injection attacks, denial of service, or compromised firmware, in different Cyber-Physical Systems (CPSs) domains. This is increasingly important in industrial IoT infrastructures, due to serious and wide impact as shown in Figure 3; especially today, with modern IoT multi-core integrated devices that provide rich functionalities but also wide attack surface, including device, network and cloud. Intrusion detection methods are mostly dedicated to ensuring network communication security [39], but these methods are undermined from IoT device heterogeneity and the highly dynamic threat landscape against them.



FIGURE 3. Impact dimensions of threat model to IoT infrastructures.

Machine learning approaches do not rely on domainspecific knowledge, but they usually require a large quantity of labeled data through, for instance, classification-based methods [50]. An inherent requirement to guarantee tamperresistant CPSs involves specification of accurate adversary models, that is, (i) a complete specification of all known attack vectors including risk assessment of identified attacks (i.e., the likelihood of the attack and the impact of exploiting each threat on the system) and, (ii) maintaining of this attack database current. On the contrary, the key advantage of behavior-based approaches based on unsupervised techniques is that they do not focus on something specific. Although these approaches can be susceptible to false positives, they are independent from any past knowledge of attack methods and their impact. However, current anomaly-based detection systems can hardly detect new types of attacks, because they are designed either for specific applications, or for limited environments. Thus, the defense capability of existing security mechanisms can be mediocre, and for instance, limited to specific distributed denial-of-service (DDoS) attacks. As these attacks can spitefully diversify the underlying protocol or the operation method, the fundamental features of DDoS attacks should become the basis of any detection method. Several research works focus on detecting the attacks by using machine learning techniques [17], [40]

and in IoT infrastructures [18]. Nonetheless, it is challenging to match the most successful detection technique to the attributes of the attack surface of IoT infrastructures in a hollistic way, or in a specific-optimized way. In this context, researchers have showed how to use even twenty-three features to detect DDoS attacks using various ML classifiers [51].

In addition to cyber-physical attacks, as it is not easy to distinguish the cause of such an abnormal situation in a given system, either a fault or an attack, detection and prevention techniques should consider both interchangeably. Faults are an abnormal state which might lead to errors or failure of the system, including permanent, transient, or intermittent, raising important concerns and defenses in industry [52], automotive [53] and medical [54] domains. Physical defects in the sensors, inside the chip, or concurrent attacks to the IoT device can be a major cause of damage to the system.

IoT infrastructures are exploding in industry, automotive, healthcare, while integrating different networks and different devices which makes it a nearly unreachable target to learn anomalous data that cause physical damage in view of unknown attack vectors and attack techniques. Due to the lack of anomalous data collections for training, ML-based detectors can hardly provide high accuracy with an adverse effect involving false alarms. Thus, there are limits to the ability to generate anomalous data such as car accident data and Cyber-physical System (CPS) faults in medicine.

Challenges for embedded devices in IoT infrastructures mainly involve spatial and temporal relationships, devices and data heterogeneity and labeled data shortage. Time-series data generation, transmission (and even prediction [55], or predicting of the future timestamp while avoiding a range of anomalies such as point anomalies, contextual anomalies, and discords in time series data [56]) are widely accepted in several domains. In this scope, recurrent neural networks (RNN) have been investigated and shown superior performance for behavior modeling. Among them, the long shortterm memory (LSTM) emerged as an enhanced version of RNN for deep anomaly detection framework for sensing time-series data in Industrial IoT (IIoT) [35], while some also using multi-dimensional sensor data fusion [57]. Several intrusion and attack recognition works [35], [58] have demonstrated the efficiency of the RNN in terms of discovering anomalies in an accurate and timely way. Even though convolution neural networks (CNNs) are inpractical to capture sequential data representations, they have been leveraged for intrusion detection due to their capability to extract spatial features. By including parallelism techniques, temporal CNN has been introduced and validated to be more efficient over the CNN since temporal convolutional network (TCN) can learn windowed temporal dependencies over long spans more convincingly [59]. Additionally, TCN exhibits improved performance compared to LSTM in many sequence problems, while at the same time, RNNs design is more complex compared to TCN. Unsupervised and supervised learning (i.e., deep learning with random forest) can be combined in In principle, to defend the network, all of the ML-based approaches listed above work at the host level. However, malicious programs operating on IoT devices, as well as wireless attacks, necessitate edge security countermeasures that are strengthened with machine learning approaches. Furthermore, because the trust boundary incorporates all of these elements, solutions at the edge level should be harmonized with typical conventional protection techniques that are already in use on cloud and SCADA servers, as well as databases. These conventional protection schemes may include monitoring and logging systems, remote access and anonymization control, and smart configuration and changes management.

III. HARDWARE-ASSISTED ML FOR SECURITY

Essentially, an IoT device is mandatory to be secured through a chain of trust. This chain is developed when an IoT device boots up only if cryptographically signed code components are first executed. These software components include bootloaders, kernel and kernel extensions, all the way from bootloader to userspace. For signing the software components, a trusted entity is responsible to provide signatures by using public-key cryptography. In particular, to establish secure boot of the microcontroller, an authentic first piece of software should be locked in a flash memory region sealed from further programming and, should implement the digital signature check of the next piece of software. In addition, the keys should be stored in specialized secure hardware to prevent not only modification but also indirect or partial extraction. The ultimate objective is to ensure a root of trust, essentially meaning that the embedded system is unclonable. Trusted paths, channels and secure communication connections (e.g., via differentiated keys, two-way communication, chained certificates) between secure signing entity, trusted module and firmware components are built on the base of authentication keys and their certificates. A secure embedded system needs to satisfy all the security requirements that involve the authenticity of the running software, the confidentiality of permanently stored elements (keys and sensitive data), and run-time state integrity.

All components of the system (i.e., components in the hardware architecture), as well as executing software and networking data, should be analyzed to find aberrant behavior that suggests security violations in a computing system. Without incorporating machine-learning algorithms, the analysis and identification activities in this process pose significant obstacles. A Network-based Intrusion Detection System (NIDS) can monitor traffic and analyze packets, hosts, and service flows to look for possible attacks in this context. This procedure is divided into two parts: algorithms for implementing effective inference techniques, such

fixed-point arithmetic to allow for accelerated run-time

embedded hardware performance [63]. Additional optimiza-

tions include tensor decomposition, pruning, and mixed-

precision data representation. These improvements are

mainly designed in hardware, based on neuro-inspired archi-

Performance and energy-wise optimizations in DNN train-

tectures, on CMOS, or with emerging memories.

as machine learning, and model building for generating an attack profile and classification for determining if the examined traffic is valid or vulnerable to attack.

The following sections aim to illustrate how current research has tackled speeding up machine learning algorithms to make them suitable for IoT devices, as well as whether these specific acceleration approaches may be used for security provisioning. Figure 4 shows the blending of domains investigated in the remaining sections of this work.



FIGURE 4. Surveying of security and machine learning research methods blending hardware and software techniques in IoT infrastructures.

A. ML AND DNN ACCELERATORS

Most research works in ML and DNN accelerators focus on computer vision domain, biomedical signal processing, etc. However, these works pave the way for integrations of such accelerators also in securing IoT infrastructures, devices and networks from anomalies.

Hardware accelerators typically are integrated to boost performance of functions either in data centers or in embedded edge-AI devices. These edge devices commonly are battery powered and hence need to operate under constrained power budgets, mostly under a five Watt roofline. Even though the scale of work differs in edge-AI and data center paradigms, both follow similar pathway to provide efficient solutions. To improve computational throughput, most accelerators use optimization strategies which involve reduced precision arithmetic, or architectural-level enhancements, such as minimizing of data movement (through using in- or near-memory computing) and increased parallelism.

In this context, researchers proposed architectural extensions for DNN accelerators, Eyeriss v2 [61], by adding a hierarchical mesh network-on-chip to limit the costly all-toall communication within local clusters. When processing DNN sparse input, even in compressed form, this results in a significant increase in throughput and energy efficiency. Alternatively, to handle data sparsity in DNNs, ENVISION proposes input guard memories and guard control units and a dynamic-precision SIMD architecture providing energyprecision scalability [62].

Research efforts to provide energy and intermittence-aware DNN inference and training, developed the Neuro.ZERO architecture, which is based on adaptive high-precision

blending of f this work. ing have driven significant research towards investigating different numerical formats. This trend is due to the fact that microarchitectural operations on fixed-point and lowprecision floating-point logic (see Figure 5) are significantly more efficient in terms of area and energy than full precision

precision floating-point logic (see Figure 5) are significantly more efficient in terms of area and energy than full-precision logic (e.g., 8-bit fixed-point addition is 30x more energy efficient and 116x more area efficient than FP32 addition) [64], [65]. More recently, researchers have proposed mixed-precision format for training by using hybrid Block Floating-Point (HBFP) format, which uses 8-bit BFP for tensors in the training operations (e.g., dot products, convolutions), and FP32 for the remaining operations (e.g., activations, regularizations) [66].



FIGURE 5. Comparative energy and area cost for different precision for 45nm technology (adapted from [64], [65]).

The support of reduced precision has fueled the recent trend in integrating DNNs also to platforms that are resource and energy-constrained such as IoT devices. Different works have shown various methods that scaled down arithmetic precision to 16-bits and even to 1-bit to optimize computation performance with minimal energy consumption [62], [67]. Contrary to early models (e.g., AlexNet, VGG) which use large number of parameters and parameters proportion of the full connection layers, modern techniques have since become popular for building compact DNNs. The key idea of these techniques is mostly based on filter decomposition for images, as shown in Figure 6, and decomposition and CNNs for time-series data [68]. These DNNs, such as SqueezeNet [69] and MobileNet [70], make a perfect fit for mobile devices and for anomaly detection in IoT monitoring [68]. DNNs today have diversified in terms of shapes and sizes that vary extensively.



FIGURE 6. Different filter decomposition solutions.

Machine learning accelerators with small footprint have shown their benefits by achieving the combination of efficiency (due to the small number of target algorithms) and broad application scope [71]. In particular, an optimized neural functional unit (mostly in terms of memory management) can achieve a speedup of $117.87 \times$ and an energy reduction of $21.08 \times$ over a 128-bit 2GHz SIMD core with a normal cache hierarchy [71]. To optimize energy efficiency for mobile devices use, Deep Neural Processing Units (DNPU) have been proposed based on optimizing heterogeneous multi-core architecture for both CNNs and RNNs [72]. Depending on the attributes of each network, the memory architecture, data paths, and processing elements are optimized for each core. Additionally, custom separation of workload can give reduced off-chip memory bandwidth needed in a CNN. Regarding an RNN, extra multiplications are reduced via quantization table-based techniques. Developers also adopt a holistic design approach to provide low-power accelerators for accurate DNN prediction for power-constrained IoT and mobile devices, using a highly automated co-design methodology that incorporates insights and methodologies across the algorithm, architecture, and circuit levels [73]. Moreover, researchers have advocated balancing the architecture in terms of cost and returns for in-DRAM calculations to speed up DNN in mobile contexts [74]. By optimizing the systolic array on a DRAM die returns include 1.7 times TOPS, 3.7 times TOPS/W, and 8.6 times TOPS/mm2 improvement over a state-of-the-art mobile GPU accelerator, while the power consumption reaches at most 4.4 W. With the objective of energy-efficiency and accuracy, researchers have developed both software and equivalent hardware implementation for feature extraction engine and the Decision Tree classifier [75]. As a result, they've proved that a hardware version of the hash-based feature extraction engine uses just 5.7 percent of the energy that the software version does. Lightweight classification algorithms have been tested in IoT contexts with time-series data, promising accuracy and scalability, and outperforming the commonly used 1-nearest neighbor with dynamic temporal warping [76]. Through use of fewer parameters results in lower calculation costs, which is ideal for real-time, hardware-assisted malware detection [32], [77].

In summary, a wide spectrum of prevailing techniques have rapidly enabled a new landscape for edge computing integrating ML and DNN-assisted processing.

B. ML-BASED METHODS FOR SYSTEM-ON-CHIP PROTECTION

Broadly, to detect patterns of abnormal behavior, various methods use memory image probing and analysis at the OS level, due to flexibility and easy access. OS-level techniques can be subject to software attacks (e.g., kernel rootkits may compromise the OS-level logging system), or even hypervisor-level forensics solutions can be the attack target itself [78]. Hence, modern methods, which establish a machine learning (ML)-based offline or runtime analysis, have shifted the initial OS-level approach so that to rely exclusively on data collected directly through the hardware. The goal is to refrain from using a hypervisor or an OS, due to credibility of the provided information, tampered by an adversary.

1) ML-BASED METHODS FOR SYSTEM-ON-CHIP PROTECTION FROM HARDWARE TROJANS

Hardware producers frequently outsource multiple elements of their design and/or fabrication processes to keep up with the rising interest for IoT devices and the globalization of hardware fabrication. These methodologies allow harmful circuits, such as Technology Trojans (HTs), to be inserted into current Systems-on-Chip(SoCs) hardware, which is becoming an increasingly serious concern [79], [80]. HTs may leak encrypted information, degrade device performance or lead to total destruction. HTs are usually divided into four categories: (i) denial-of-service, (ii) function change, (iii) performancedegradation, and (iv) information leakage. For instance, DNN inference behavior can be successfully tampered at run-time with deliberately degradation of the victim inference accuracy through memory-efficient rowhammering and precise flipping of targeted bits [81].

Researchers primarily employ two fundamental detection and defense techniques, (i) tackling side-channel attacks via leaking of power/thermal/delay/optical/electromagnetic information, and (ii) logic testing-based, by using keyto-signature mechanisms and assuming the existence of a "golden model".

Early works used side-channel information for Trojan identification such as analysis of the path delay to generate a unique fingerprint that can be used to distinguish tampered chips [82]. Based on ARMv7 microprocessor's operating frequency deviations, by integrating analog Trojan circuit it is

shown to detect an extremely rare HT, triggered by successive toggling events [83]. Additionally, the measurement of process control monitors (PCMs) was combined with a machine learning technique, a one-class Support Vector Machine (SVM), to obtain a more precise categorization boundary in identifying abusive behavior of circuits, which considerably increased the efficiency and accuracy [84]. By reverse engineering (RE), it is shown that recovered images can represent the physical structures and layout of the ICs, which are classified based on support vector machines, particularly oneclass v-SVM, to distinguish between random differences and the systematic differences caused by Trojan insertion [85], as shown in Figure 7. This means that the aim is to identify Trojans while allowing for manufacturing and reverse engineering process variances. These approaches can be paired with the usage of Deep Convolutional Neural Networks with intrinsic extraction of invariant and non-linear features to overcome manual or domain-specific feature extraction [86].



FIGURE 7. Block diagram of one-class v-SVM trojan detection approach.

Moreover, hardware Trojan activation is proved to be successfully detected by comparison of power use between Trojan clear and Trojan embedded benchmarks via using machine learning techniques [87]. Also, by using random forest classifier, recent works demonstrate how to extract effective Trojan features from hardware-Trojan infected nets in ICs [88]. Following a hybrid approach, researchers propose to combine the signature extraction mechanism with machine learning algorithms to develop a self-learning framework, as depicted in Figure 8, that can detect the intruded integrated circuits [89]. As this researh work shows, the decision tree (DT) algorithm is the best among selected prediction algorithms (i.e., decision tree from eager learning algorithms, bayesian classifiers from probabilistic learning and k-nearest neighbors from lazy learning) in term of accuracy and precision.

Additionally, to tackle complex and expensive on-chip learning-based approaches, a deep invasive methodology with a lightweight, low-power ML-based monitor for HT detection can give competitive benefits, given of a proper training dataset is utilized [90]. Alternatively, by using on-chip sensors and classification on the basis of statistical distribution of grid-partitioned power consumption, runtime Trojan detection approach gives promising results [91]. In particular, in the design phase, the ML training process uses power profiles by measuring the combined power consumption of each component involved in a particular pipeline stage along with the Trust-Hub benchmarks [92], which are then used for HT detection at runtime.

With an actually realized hardware architecture for Support Vector Machine kernel, a proposed security framework gives



FIGURE 8. ML-based trojan detection methodology by using process and mismatch variations as timing signatures (adapted from [89]).

a detection accuracy of up to 97% for three expected Trojan attacks for a NoC-based many-core architecture [93]. The detection efficiency, in terms of accuracy (without ignoring the complexity and integration convenience), depends both on the type of the Trojan attack and the type of the machine learning model used. Given a supervised learning model, such as SVM, DT or LR, traffic diversion attacks can be detected with an accuracy that exceeds 95%. For example, by using decision trees, core address spoofing, route looping and traffic diversion can be detected with an accuracy reaching 94%, 95% and 99%, respectively [93]. In contrast, in this category of attacks, the unsupervised learning models prove to be more deficient in terms of prediction accuracy. Figure 9 summarizes key points regarding classification of discussed strategies.

It must be noted though, that there is still lack of machine learning-based algorithms for identifying the HTs compared to detecting HTs. However, the usage of classification methodology which involves machine learning for HT detection is complex and depend on the detection techniques used (e.g., shallow ML algorithms for detection are mostly targetspecific and prone to underfitting or overfitting).

2) DETECTION AND PROTECTION AGAINST ATTACKS TO ML COMPUTING

An additional direction of research involves detection, at runtime, of the correctness of a neural network's computations, such as Safe-TPU [94]. This is essentially a verifiable Trojan resilient hardware accelerator for DNNs that detects arbitrary Trojan misbehaviour, regardless of how the Trojan is designed or triggered (time-based or cheat-code based Trojans). Essentially, besides the software attacks, hardware trojans might be carefully designed to compromise the neural network's integrity, in terms of the trigger, or of the payload (i.e., the input, computational block, intermediate data and output) [38]. Modern object detection platforms, such as YOLO [95] and Mini-YOLOv3 [96] for embedded devices, expose a hardware attack surface, as shown in Figure 10, with a number of options, including:

- Model Corruption: compromise the model parameters stored in memory so that the model results deviate in all tasks
- Backdoor Insertion: alter the model itself which is stored in memory so that it provides near random results partially or fully



FIGURE 9. Taxonomy of HT attack detection by ML-based methods.



FIGURE 10. Attack surface of a YOLO object detection framework, composed of 24 convolutional layers, followed by two fully connected layers.

- Model Extraction: extract the model from the device during run-time or via proving non-volatile memory
- Spoofing: interfere and manipulate the model input data through tampering with the input sensors or with the environment
- Information Extraction: infer model information by capturing and analyzing the physical side-channels

To enable accurate NN Trojan detection on resourceconstrained embedded devices, recent research efforts target algorithm/hardware co-design for an end-to-end method through using a pair of input (based on Discrete Cosine Transform, DCT, extraction) and latent feature analyzers [97]. To provide strong integrity and privacy guarantees for a NN execution, authors used secure enclaves, i.e., a Trusted Execution Environment (TEE) and at the same time outsource non-critical functions from a TEE to a faster co-processor [98]. Neuron obfuscation can effectively combat increasing risks to IoT edge devices and enable security of critical data and DL model parameters, while relying on a secure key storage facility supplied by a hardware rootof-trust such as Trusted Platform Module (TPM) [99]. In IoT environment, it is important to determine adversarial attacks in real-time, attempting to compromise Network Intrusion Detection Systems (NIDS) that employ DNNs and CNNs for identifying benign from malicious network traffic [100]. In this scope, designing accelerating circuitry is an emerging topic in the deep neural networks area for security, by for example, using memristor crossbar arrays to significantly improve the throughput of the visual adversarial perturbation system [101].

C. ML-BASED METHODS FOR EMBEDDED SoC PROTECTION FROM MALICIOUS SOFTWARE

Most techniques that involve ML in device level are mainly custom specific, in the scope of the type of attack surface and of the device attributes. For instance, an ML-based approach is proposed in wireless networks-on-chip (NoCs), to identify jamming-based DoS attacks and evesdropping originating from either an internal or an external attacker [102]. They use burst error correction codes to estimate the number of burst errors in packets captured at the receive transciever. With the aid of ML classifiers (artificial neural network, support vector machine, k-nearest neighbors, and decision tree), DoS attacks are then distinguished from random transient burst errors (due to power fluctuations, ground bounce or crosstalk) and a defense unit is notified.

To protect embedded devices from malicious software components that can perform hijacking attacks in the control flow, such as code-reuse² attacks (e.g., like buffer overflows, return- or jump-oriented attacks) [103], designers' trend involves control flow integrity (CFI) checking. CFI examines the code execution flow graph in traces of various granularities and attests to the validity of these valid execution traces in general. However, in real-time embedded systems, especially those with restricted resources, hardwarebased techniques are being developed for efficiency and resilience to software assaults, allowing for a novel way to resisting malicious software. By using the ARM CoreSight module in an ARM-based IoT environment, recent work proposes a hardware-based workload forensics framework for IoT systems [104]. By recording the spatial and temporal architecture of the address space they create a workload identification scheme that combines numerous machine learning algorithms (such as the Long Short-Term Memory (LSTM)-Recurrent Neural Network (RNN)) to assess and comprehend the workload being executed at the granularity of a process in real time. To realize anomaly detection pre-learned thresholds form the basis of comparison with the classifier outcome and thus potentially illegal program behavior is filtered out.

²Software anomaly caused by illegal or unintended redirection of the logical program flow, to instructions already present in memory.

In general, common techniques for detecting malware or side-channel assaults rely on the use of hardware performance counters (HPC) and machine learning algorithms to build a model of the program's behavior. HPCs in multi-threaded processors are monitored in real-time to detect abnormal activity. A classifier is used to detect out-of-profile behavior by comparing retrieved characteristics to features from a previously set baseline [105]. To characterize application behavior, alternative techniques collect low-level architectural information such as profiling data from memory address references, instruction opcodes, and Translation Lookaside Buffer (TLB). However, because they rely significantly on the determinism, authenticity, accuracy, and availability of the information leveraged by hardware and software performance counters, even ML-based generated models may increase the already broad attack vector surface. Performance counters may unintentionally degrade the performance of machine learning classifiers because of data polution. All techniques presume the application's training phase is reliable, which is a prerequisite of most behavior-based intrusion detection systems. The runtime monitoring entity is therefore expected to be trustworthy and untampered with.

Another approach uses the inspection and analysis of electro-magnetic (EM) side channels to classify the kind of operations performed on a processor and so identify software execution sequences with no need to instrument the program; this information may be utilized for anomaly identification [106]. Aternative methods aim to decompose the time series to small and interpretable components, or to characterize the EM leakage of electronic devices via Fast Fourier Transform (FFT) and identify the frequencies that represent critical part of the executing program [107]. However, these techniques have drawbacks largely due to sensor noise and measurements sensitivity. Such methods for analyzing and evaluating a device's side-channel security via leakage detection, as well as standards (such as ISO/IEC 17825:2016) that provide a systematic set of leakage detection tests, have been observed to produce false positives [108]. Furthermore, finding accurate and suitable features and selecting effective parameters among many features is a difficult topic for ML to use for a high detection rate. Prior to implementing any security mechanism, one essential necessity is to eliminate any link that may transport trustworthy information from a secure region to the outside world, as this poses a risk [109].

IV. ML-BASED SECURITY IN EDGE IOT DEVICES

Modern embedded systems inside IoT infrastructures necessitate a higher degree of dependability, accessibility, and robustness, for industrial, automotive and healthcare applications. Because traditional machine-learning approaches that run in the cloud cause reaction time delays, current innovations suggest that ML techniques and smaller-scale models will increasingly shift to edge devices, in the proximity of data sources. Big data transfers to cloud-hosted machine learning processing may cause networking flooding and large round-trip latency as compared to edge processing. Meanwhile, millions of low-cost tiny computational devices in the real world represent a significant amount of underutilized processing power. Some learning algorithms, such as instance-based learning, may, however, be too costly for edge devices. As a result, the accuracy of outcomes in IoT end-nodes may not be as great as in cloud-based systems in some circumstances.

A. ML-BASED INTRUSION DETECTION AND PROTECTION IN IOT DECENTRALIZED ENVIRONMENTS

To tackle security and privacy issues, several approaches use ML-based techniques integrated in schemes spanning end user-fog-cloud environments. Whilst conventional cloud computing solutions might be adapted to handle some security and privacy concerns with fog computing, the latter's unique features, such as decentralized infrastructure, mobility support, location awareness, and low latency, provide unique security and privacy challenges. Because of the decentralized architecture of fog computing, it is difficult to collect and manage evidence and behavior information about fog nodes to evaluate their trustworthiness and build a trust evaluation model for all fog nodes in the network, behavior-based ML methods for increasing security and privacy in fog environments are difficult to achieve.

Fog nodes that are semi-trusted are responsible to realize a trustworthy framework to aggregate multiple sensors that is based on machine learning [57]. To alleviate cloud-based overheads, the proposed technique uses a trained model to forecast the contribution of sensor readings to the aggregate sum. Additionally, to protect the training dataset against differential assaults, this technology uses differential privacy (e.g., via introducing noise).

In a different perspective, contrary to intrusion-detection schemes defending a single domain in traditional networks (e.g., enterprise, cloud, business domain), recent strategies employ learning from various domains to identify various attacks [110]. The edge data collector is responsible for collecting the IoT data, while the edge analyzer is responsible for analyzing collected data and IoT device behavior and, the edge controller, which is based on software defined networking (SDN), is responsible for gateway configuration.

Therefore, to both optimize response time and resilience of fog layer (see Figure 11), researchers propose various orchestration techniques [111], [112], or employ machine learning-based methods in a secure-conscious manner, such as the MAPE-K model [113]. This model contains four main components: management, analysis, planning and execution. Aggregated data are partitioned and packetized depending on the data type generated from sensors, and communicated via using 128-bit AES-CCM encryption. On the basis of the type of these produced and collected data, training is performed at the cloud server and the outcome model is then executed at the edge device. However, the ML algorithm that is used to generate the model must respect the edge-device constraints.

Anomaly detection schemes tailored for IoT cybersecurity have also been presented through using IoT gateways to host



FIGURE 11. Improving response time in hierarchical fog-assisted computing architecture through mapping and moving functions and data to the fog layer.

an artificial neural network [114]. Such techniques can effectively determine correct and incorrect delay and sensor values via three-input neurons. As identified, the main challenges for anomaly detection in IoT data are quantity and heterogeneity. In the same scope, a deep recurrent neural networkbased malware detection methodology for the ARM-based IoT applications has provided promising results [115] via analyzing IoT devices application opcodes. By implementing three different long short-term memory configurations, this research approach showed 98.18% accuracy to detect malware with respect to the tested data set. Additionally, to improve the trustworthiness of services in a decentralized IoT environment, researchers proposed a reinforcement learning (RL), RL-based approach to determine the service resource allocation scheme in different time periods [116]. In all aspects of cybersecurity, by adopting a data-driven approach, anomaly detection algorithms prove to provide a valuable effective toolset. Most machine learning-based IoT approaches for malware hunting focus on energy consumption patterns [117] and application's opcodes [118].

B. HARDWARE-ASSISTED ML IN IOT DEVICES

Devices that incorparate ML for detecting and subverting attacks commonly adopt software-based solutions, such as anti-virus applications. These solutions, though, are susceptible to high risk; sophisticated malware may be equipped with smart deviation capabilities such as obfuscation, which may be successful since traditional protection schemes mostly rely on matching patterns and signatures. The tamper-immune hardware metrics prove to be an improved security feature compared to the high-level software metrics, since software features can be jeopardized via obfuscation. Hardwareassisted ML semantically involves different methods and architectures categorized as follows.

- Hardware assistance for making ML detection more accurate, i.e., minimize false positives
- Hardware accelerators to build faster ML models and inference engines

Machine-learning-aware and deep-learning-aware optimizations of processors (i.e., vector width improvements, SIMD instructions parallelism, low-precision FP computations) to boost the performance of a range of deep-learning applications

To optimize ML-based malware detection accuracy, recent research works propose real-time collection and analysis of hardware traces [119]. These hardware-supported instrumentation traces include (i) embedded trace buffers to collect functional values of a number of trace signals over a time window in clock cycles granularity, (ii) hardware performance counters to determine statistical behavior in terms of specific architectural features such as bus or memory accesses, cache misses, branch prediction, and (iii) Network-on-Chip (NoC) traffic to provide insight in communication patterns. Experimental results show that machine learning can be effective in malware detection by utilizing such hardware traces. In a different perspective, through using architectureagnostic methods for forensics analysis, researchers propose to reconstruct executed workload at the granularity of a single process by using the extracted features, through minimal information obtained from the processor's translation lookaside buffer [120]. Alternatively, by exploiting hardware performance counters to collect fine-grained data for each system call of unknown programs, these unknown programs can be categorized into benign or malicious [121], [122]. The programs behavior can vary, with a significant trace comprising of thousands of system calls, while some have a short trace limited to less than a hundred system calls. To tackle such variations, captured performance counter data are reduced to a uniform dimension and then classified via decision trees, random forest, neural networks, adaboost, k-nearest neighbors with promising results in a range of fidelity [121], [123]. Even more aggresively, others introduce the use of dedicated on-chip learning controllers to perform the analysis directly in hardware, possibly even in real-time, for instance by embedding neural network or logistic regression prediction co-processor to decide based on instructions and memory access extracted features [124]. Such approaches require specialized hardware designs, but offer a low power consumption footprint with zero software interference. Since most recent processors for IoT devices are equipped with hardware performance counters that can be used for malware detection, inexpensive methods can be employed (via using low-level hardware events) to detect threatful alterations in the firmware of embedded control systems [34]. By exploiting an augmented number of hardware performance counters with reduced accuracy, limited added value is shown for different hardware classifiers to achieve better performance, accuracy against area overhead, while the combination of classification algorithms has a good performance outcome [125]. In summary, a growing interest involves the usage of low-level microarchitectural features collected from processor's performance counter registers to implement hardware classifiers for malware discovery, with little concern of combining higher level behavior

(i.e., such as operating system or network activity). This strategy offers isolation from software threats at the risk to miss new, sophisticated threats. ML-based detection models that use HPC-based approaches need to become robust against algorithm subversion attacks, especially when securing Post Quantum Cryptography (PQC) implementation on resourceconstrained devices, a key requirement to maintain their integrity [126].

Today, researchers suggest relocating a classifier algorithm (such as DT) in hardware to enhance both the energy efficiency of anomaly-based intrusion detection systems for probing assaults and the restricted throughput of software in resource-constrained edge devices [127], [128]. The isolation from the software environment and intrinsic robustness of circuitry against tampering are two further advantages of mapping an ML algorithm in hardware. After contrasting several approaches (e.g., naive bayes, support vector machine, k-nearest neighbor, random forest, and artificial neural networks) for real-time performance, hardware-based classifiers demonstrate excellent performance [129], with random forest outperforming other algorithms with a maximum accuracy of 98.5 percent [130]. A full framework for deploying CNN on embedded systems has also been described, which uses a mixed pruning strategy to compress CNN models and thereby alleviate memory and performance issues [131]. However, most FPGA implementations exhibit high cost in power consumption, with some exceptions [127], which does not allow integration with microcontrollers and resource-constrainted IoT devices. Earlier, solutions presented also developing feature extraction module in hardware and the use of principal component analysis as an outlier detection method for NIDSs with detection rates exceeding 99% [132].

Towards dedicated, specialized AI-workload processors, BrainChips's Akida neuromorphic processor is a revolutionary advanced neural networking processor that brings artificial intelligence to the edge [133]. The Akida NSoC is designed for use as a stand-alone embedded accelerator or as a co-processor, while also including interfaces for ADAS sensors, audio sensors, and other IoT sensors. Moreover, NeuroEdges are devices that support the implementation of edge computing systems using neuromorphic chips, named NM500 [134], and common commercial embedded boards, however mostly targetted to face recognition [135]. Research results demonstrate considerable advantage for real-time computations, thus savings in terms of the burden of requiring many datasets for effective training.

In the scope of bringing AI at the edge, IoT devices are also emerging with hardware support. Recently, ARM introduced enhancements towards boosting ML processing on top of the ARM Cortex-M55 processor, that can be up to 15 times faster than the previous version, and ARM Ethos-U55 NPU, the first micro- Neural Processing Unit, micro-NPU, for Cortex-M architecture, which can speed up ML performance by up to 480 times [136]. By integrating Deep Learning Accelerator (DLA), NVIDIA DRIVE AGX Xavier can deliver an incredible 30 TOPS for automated driving [137]. To enable real-time sensing with limited energy generated by energy harvesting, Renesas embedded AI (e-AI) [138] demonstrated power efficiency of 8.8 TOPS/W [139]. The Renesas accelerator developed a processing-in-memory (PIM) architecture, an increasingly popular approach for AI technology, in which multiply-and-accumulate operations are performed in the memory circuit as data is read out from that memory.

In addition to emerging devices with AI-oriented hardware extensions, modern ML tools oriented to help in running AI algorithms on microcontrollers, facilitate inferencing based on models trained with TensorFlow, Keras, PyTorch, Caffe and others [140], [141]. The application code can directly use these kernels to realize neural network models on ARM Cortex-M CPUs. Moreover, developers deliver microcontroller optimized libraries, such that neural network inference can achieve 4.6X improvement in runtime/throughput and 4.9X improvement in energy efficiency [142]. First, these optimized functions accelerate key neural network layers, such as convolution, pooling and activations. Second, the optimizations aim to reduce the memory footprint, which is key for memory-constrained microcontrollers. Alternatively, these kernels can be used as primitives by machine learning frameworks to deploy trained models.

STM32Cube also helps with the easy integration of standard AI algorithms in microcontrollers. Automatic conversion of pre-trained neural networks and integration of the resultant optimized library into the user's project are made possible by the particular AI ecosystem. Cube.AI tool offers not only mappping a neural network on an STM32 MCU but also optimizations. For instance, the code generator opts for folding some of its layers and reducing its memory footprint. In particular, to optimize for condition monitoring and anomaly detection, and hence reducing anomaly detection time, STM's FP-AI-NANOEDG1 manages sensor input data collection, on-device learning sessions and inference models in real-time [143], [144]. These tools claim to make it easier to create machine learning libraries that include both inference and edge training. The purpose is twofold. First, predictive maintenance is seamlessly enabled. Second, for assault detection, sensor patterns are used in a self-learning, simplified method. The requirement for extensive knowledge in machine learning, data science, or developing neural network models is becoming obsolete as a result of tool automation. At the same time, FP-AI-NANOEDG1 offers coverage of the entire development of the machine learning cycle. This means, it helps from the data set acquisition up to generating libraries by the NanoEdge AI and integrating the application on the physical node, as well as the security and detection with sensor patterns self-learning and self-understanding. Essentially, an STM32L4R9ZI ultra-low-power microcontroller supports all tasks, data collection, learning session and real time inference, while processing physical sensor data as input.

Advancing both signal processing and neural network applications to edge-devices are also emerging for new embedded platforms that integrate multiple cores in parallel,

such as GreenWaves Technologies GAP-8 and GAP-9 (i.e., nine RISC-V cores) [145], to enable embedded machine learning in battery-operated IoT sensors, mainly focusing image processing domain. Such systems-on-chip (SoCs) are among the most advanced low-power edge nodes available in the market, embodying the PULP architectural paradigm with DSP-enhanced RISC-V cores, while frameworks have been developed exploiting SoCs features, such as hardware loops, post-modified access LD/ST, and SIMD instructions down to 8-bit vector operands [146], [147]. Additionally, to provide agility for a variety of different neural network techniques, a novel domain-specific Instruction Set Architecture (ISA) for NN accelerators, called Cambricon, has been proposed [148]. This is a load-store architecture that integrates scalar, vector, matrix, logical, data transfer, and control instructions, based on a comprehensive analysis of existing NN techniques. In summary, Figure 12 gives an overview of different directions in bringing ML processing at the edge for efficient data processing in secure manner.

In conclusion, hardware specialization is a popular approach to accelerate the computation of neural networkbased applications. Besides neural network and deep learning (DL) accelerators, specific microarchitectural techniques and even software methods succesfully present high performance and attempt to save energy. Microcontrollers can also provide hardware and software support for low-precision computing [149]. Research results via lower precision fixed-point arithmetic are promising in terms of memory footprint, inference time and power efficiency (by using TensorFlow Lite for microcontrollers, STM32Cube.AI and a custom tool MicroAI) [150]. DNNs also perform computations with other patterns, such as sparse lookups, vector operations and deconvolution [151]. Future CPUs will also host dedicated DL accelerators to accelerate not only such operations but also crypto- and analysis functions, thus, bringing all worlds together to resource limited devices. Vendor-optimized libraries will remain essential to leverage all the performance capacity from a processor.

However, most of these machine learning solutions mostly focus on sensor data fusion and help ecosystem to advance the future of automotives, smart buildings and wearable computing, but rarely consider anomalous components behavior in an IoT infrastructure as a prime goal. On the other hand, embedded hardware security in IoT infrastructures is a necessity to protect the identity of devices, to secure the trusted execution of their applications against tampering, and to protect the privacy and security of data they generate. Nevertheless, protection techniques such as HSM and TPM enhancing hardware security are scarcely linked to ML and DNN methods in this scope.

In addition, while a variety of useful mechanisms to protecting against memory vulnerabilities at run time have been presented, such as fine-grained tagged memory systems [152], support of pointer authenticity [153], and hardware-assisted scope enforcement [154], they are seldom integrated with ML-based solutions.

H/W-assisted Methods	Advantages	Drawbacks	Comments	
Z. Pan et al., [119]. Use HPC, K-nearest neighbor (KNN), Random forest(RF), Decision tree(DT) and Neural Networks(NN)	Real-world benchmarks	Architecture- specific	HPC limitations: • Training data spoofing, non-determinism,	Hardware Performance Counters
S. Kadiyala et al., [121]. Fine grain extraction of HPC, PCA-, RF- based dimensionality reduction, KNN, RF, DT, NN, AdaBoost. KNN	Real-embedded CPU(Cortex-A9)	Data spoofing, HPC counters manipulation	over-counting [123] sensitive when using PQC signatures [126]	
H. Wang et al., [122]. Two-phase runtime detection (OneR, MLP (multilayer perceptron), J48, and BayesNet ML algorithms) and identification (BayesNet classifier), 2 features: L3 Misses, L2 Hits	Identify known & unknown attacks	Accuracy vs moni- toring overhead, limited # of HPCs	Suitable for runtime detection of side-channel attacks for IoT devices	
B. Zhou et al., [123]. Show low robustness of HPC-based malware detection, by using 6 low level features with DT, RF, MLP, KNN, AB, NB to detect malware on x86	Use F1-score, extensive experi- mental results	Limited analysis on embedded malware, not real- world samples	Risk for malware crafted to perturb HPC patterns look similar to benign application patterns	
Ozsoy et al., [124]. Use Logistic Regression (LR) and ANN for malware detection, augmenting processor pipeline H/W detector, features: instructions, memory patterns, architectural events	Real-time H/W detection	Overhead and complexity	In-built ML in microproce- ssor enables low power consumption footprint and zero software interference	
Sayadi et al., [125]. ensemble learning classifiers to boost the performance-robustness of general ML classifiers, via use of 8 ML classifiers, extensive bagging, adaptive boosting, aggregation	Remove need for >8 HPCs, long training	Computation over- heads no fit for light IoT devices	Extend to defending ML- classifier parameter modification attacks	
L. Zhou et al., [120]. H/W event monitor on TLB miss events & related instructions, KNN, SVM, RNN-LSTM (for process classifi- cation), probability estimates, auto-encoder (for outlier detection)	Hardware- agnostic	Hardware- invasive	Combining of features, granularity vs feature space - accuracy of process iden- tification &outlier detection	H/W Events
A. Franca, et al., [127]. H/W decision tree algorithm to detect probe attacks from benchmarks: attacks: denial of service (DoS), probing, remote-to-local (R2L), and user-to-root	Energy savings 0.03% vs S/W, throughput 15.4x	Not integrated in a full IoT system for IDS	NSL-KDD dataset using 9 manually selected features out of 41; 96.5% accuracy on the train set and 77.8% on the test set	si
[128] [129] [130]. Investigation of Decision tree, k-NN, random forest-based network classifier on FPGAs, with traffic traces from naïve Bayes, linear SVM, polynomial SVM, KNN, random forest, and artificial neural networks	RF on FPGA: 96.5% accuracy, 0.834 F-score	Deep Learning usually superior but needs more resources	Speedup by hardware acceleration vs software reaches 92.64 and 47.68 on UNIBS & UNB datasets	Accelerato
X. Chang, et al., [131]. Filter pruning to remove unimportant kerm portion of feature maps and lowers the burden on memory, data bi 8, input segmentation, memory reuse to reduce the memory footpr	FPGA accelerator for NN in embedded devices	H/W		
A. Das, et al., [132]. Feature extraction module (FEM) and PCA a for live intrusion and anomaly detection	FPGA accelerator for FEM and PCA, network-based			
[133]. BrainChip: Mesh of Neural processors on Chip, event domain convolution neural processor or fully connected neural processor (based on spiking neural networks, SNNs), event- based: capable of learning without re-training	Ultra low power Edge Al	Extensive investigation of robust SNNs in cybersecurity	Programmable weight bit- precision for throughput vs accuracy, protected in em- bedded SRAM in each NPU	
[134] NM500: Neuro-memory architecture, neurons assignment to different contexts or sub-network allows building hierarchical or parallel decision trees, Restricted Coulomb Energy (RCE) classifier or kNN	NeuroMem delivers GigaOps while run- ning at clock freq in the order of Mhz	Limited focus on cybersecurity	Open-source libraries sup- porting NeuroMem, can learn, recognize patterns, save/restore the knowledge	cessors
[136]. ARM Ethos-U55, microNPU: CNNs and RNNs and future LSTM, RNN, pooling, activation functions. Supports Int-8 and Int- 16: lower precision for classification and detection tasks	480x uplift in ML performance over existing Cortex-M	Limited focus on cybersecurity	Wide S/W support, TensorFlow Lite Micro Runtime, CMSIS-NN, Optimizer	d Al Proc
[138]. Renesas Al accelerator called DRP-Al contains Dynamically Reconfigurable Processor(DRP) and Multiply- Accumulate Calculator for Al(Al-MAC), PIM architecture	power efficiency of 8.8 TOPS/W	Inference	DRP-AI performs AI infere- nce by co-working withDRP and AI-MAC, 16b-floats	ecialize
[144]. STM LSM6DSOX, INEMO, ML-processing assisting MCUs, 8 parallel decision trees, FP-AI-NANOEDG1 covers all ML cycle for AI on MCUs	Ultra low power in IoT MCUs	Limited focus on ML-based cybersecurity	STM32L4R9ZI ultra-low- power combines for security	Sp.
[145]. GAP-8, GAP-9 handles sophisticated neural networks such as MobileNet V1, 8-, 16-, and 32-bit precision, vectorization of fp- and fixed-point arithmetic	Low power with 50 GOPS	Limited focus on ML-based cybersecurity	806µW/frame/second (Al- vision), open-source SDK for GAP IoT Apps	

FIGURE 12. Classification and comparison of hardware-assisted ML-methods towards IoT embedded processing and security.

C. SECURE ML AND DL INFERENCING USING TRUSTED EXECUTION ENVIRONMENTS

Different methods guarantee the secrecy of the assets engaged in ML computations in untrusted computing environments to ensure resilience of ML models and derived services against malicious actors. The protected assets can comprise data, machine learning models, and computation results that can compromise the confidentiality of the protected assets indirectly. In this context, researchers suggested employing trusted enclaves to offer data integrity inference, or applying privacy-preserving algorithms with the help of ARM TrustZone to safeguard peripheral access for ML data privacy [155]. Thus, even if the adversary has complete control over the software running in the user's device's normal world, including privileged software like the commodity OS, the enclave with the ML model is attested by a SANCTUARY core [156], which creates and securely stores a cryptographic hash of the enclave's initial memory content.

Designs for DL model calculations employing tiny TCB size and limited secure memory are demonstrated for mobile and IoT devices by using the benefits of TEEs and appropriate device accelerators [157]. The supplier is required to present the genuine DL model's cryptographic hash to authenticate the DL model's integrity. Sensor data is also safely fed into the TEE using secured drivers, while encrypted data is decrypted inside TEE and sent to a protected accelerator at

each stage of the inference process (e.g., a GPU), which is splitted to minimize the code base inside the trusted enclave [157], [158]. Similarly, only the most vulnerable layers of a DNN are concealed inside the TEE to prevent inference attacks [159].

TEEs have memory constraints, and the move from the untrusted domain to the TEE adds overhead, as shown in research that assessed TEE performance characteristics proving TEE-based functionality to be expensive to both invoke and execute [160], [161]. However, various promising appoaches offer significant benefits for ensuring trusted ML model execution, such as cancelling inherent memory limitations thus allowing to securely run complex models [155], or, providing better performance for protecting ML services [98], or, removing the software layer to enable a secure OS and enabling more trusted applications to run at once [162].

V. DISCUSSION

This section presents key points and challenges regarding cross-cutting directions as surveyed in prior sections.

A. CHALLENGES FOR REALIZING ML WITH HARDWARE SUPPORT AT THE EDGE

From the perspective of the hardware architecture for DNN and machine learning, modern realizations are emerging but are mostly application-oriented. In terms of inferencing, no one architecture appears to stand out in terms of delivering critical machine learning hardware primitives to serve a wide range of applications, particularly at the edge. The field of machine learning is still in its early stages, while promisingly inferencing is shown to perform on the microcontroller by a variety of embedded systems. Provided light ML is satisfactory, such as in keyword spotting, or use-cases where response time is not critical, such as analyzing offline photos, then the microcontroller is capable of performing at such scale. A promising solution involves the realization of ML-based malware classifiers in microprocessor hardware with significantly reduced overhead as compared to the traditional software-based methods [32].

Despite the fact that early innovations utilised GPUs, which enabled a big leap forward in AI capabilities, power consumption is an important consideration in IoT devices. Inferencing via using Tensorflow models on mobile devices consumes more than the half of the consumed energy (57%) for data movement [163]. Hence, researchers propose new architectures, mostly based on processing-in-memory organizations (PIM) and small fixed-function accelerators (PIM accelerators), such as data packing and quantization to make machine learning inferencing more energy-efficient [164].

With regard to ML for secure IoT infrastructures, most hardware-oriented works mostly use Snort rules [165] rather than ML-based anomaly detection; advancements though recently demonstrated feature extraction algorithms which are suitable for hardware implementation and promising results of feature selection methods with two simultaneous objectives, accuracy and energy consumption [75]. Additionally, for different cyberphysical systems, different compatible neural network architectures should be adopted [17]. Further, ML could play a significant role in enabling asymmetric elastic cryptography in IoT but there are challenges that need to be addressed [166], such as IoT-based anomaly datasets, probability and exact threat identification, authentication of the training data sets, zero-day attacks and real-time firmware updating of millions of devices.

B. ML-BASED SECURE PROCESSING IN HIERARCHICAL IOT ENVIRONMENT

If workloads become heavier (e.g., big data in industrial applications, biomedical imaging, genomic systems), and where performance is critical or power efficiency is a concern, then different solutions appear for IoT resourceconstrained devices, ranging from microarchitectural support for AI (i.e., at instruction level [140] and accelerators [143]), to fog-oriented solutions for ML-based applications and anomally detection systems. Fog computing, as a rising computing paradigm along with SDN and NFV technologies, can become a powerful solution in securing a variety of connected industrial environments [167]. Despite the abundance of huge data in the vicinity of loT, creating and deploying strong attack detection systems for loT devices is difficult due to resource limits, latency sensitivity, and distribution concerns [168]. As fog computing provides a distributed environment with multiple fog nodes near to IoT devices in the edge layer, recent methodologies have demonstrated the usefulness of LSTM-based DL models in cybersecurity to identify a variety of threats with high detection and accuracy rates [169]. However, implementing a heavy DL detection solution directly on low-capacity IoT devices (to detect even morphing attacks), detecting multiple threats with high detection rates and accuracy rates, and monitoring and updating the detection system to identify new attacks remain difficult.

In the scope of improving ML-based IDS at IoT system level, in particular to address the strict latency requirements that challenge the detection of cyber-attacks, alternative proposals include a fog architecture to benefit from the low latency provided by fog nodes [31]. Further, hardware support for ML inferencing is deemed important for real-time IDS methods. However, as researchers show [170], program behaviors tend to deviate at an early stage of their execution and may therefore be benefited to perform the real-time monitoring and identification analysis using hardware techniques as well [170], [171].

C. METHODS AND TOOLS SUPPORT FOR REALIZING AUTOMATED AND TRUSTED ML-BASED SECURITY FOR IOT DEVICES

As the EDA tools and methods keep evolving, the implementation of hardware IDS methods in IoT devices on the basis of machine-learning algorithms and even with Trojan security aware methods [172], are increasingly boosted. This is facilitated by the wider acceptance of continuously more efficient high-level synthesis tools, based on widely known and used languages, such as OpenCL, C/C++, or MATLAB, thus enabling software designers to take advantage of FPGA technology [129], [173]–[175]. On top, with the goal to provide complete end-to-end toolchain to empower domain scientists to design machine learning algorithms for lowpower devices, new developments are presented for a range of devices [176].

Additionally, in adversarial environments which are inherently non-stationary, such as the cyber security domain, ML/AI-based IDS methods and security critical applications require further advancements in terms of reliability to address adversarial machine learning (e.g., machine learning poisoning of training data-sets and attack models) with degrading sub-optimal decisions, thus resulting in endless cyber-war gaming between defense and attack strategies [26], [49], [177], [178]. Finally, an orthogonal line of research should pursue protection of both IoT device and system secrets, even in the presence of compromised system software layers and malware.

IDS and defences developed to protect from adversarial examples, have shown great accuracy through employing DNN methods, but also a wide space for parameters tuning and reduced robustness due to adversaries capacity to evade even if an adversary is oblivious to a specific defense [179]. Essentially, an important challenge today involves not only designing accurate DNN-based defense schemes, but additionally producing interpretable results in terms of understanding how the ML algorithms reach into the conclusion for detecting attacks. Complementary to efforts towards mitigating false classification by augmenting training data for compensating undertraining, new schemes propose increasing quality of explanations for individual classification outcomes for security applications [180], thus raising the trust of users.

Furthermore, the defense strategy should advance its understanding of the pathway and parameters to generate a non-binary detect decision, jointly with how an attacker might react to any defense. The protection scheme needs to ensure that the defense remains secure against an attacker who discovers how the defense works.

VI. CONCLUSION

Machine learning and edge computing solutions promise to efficiently distribute the processing needs across devices, servers, and gateways so they can act on sensors data from heterogenous devices in real time and predict outcomes locally. It is also widely recognized that employing machine learning and neural network-based methodology is able to overcome quantity and heterogeneity challenges of IoT devices and data in detecting anomalies in the data sent from edge devices (through for example focusing on behavior and protocols). However, machine learning and deep learning algorithms are generally computationally and memory intensive, making them unsuitable for resource-constrained environments such as IoT, mobile, devices and gateways. To efficiently implement these compute and memoryintensive algorithms within the IoT computing space, especially in terms of energy requirements [75], innovative optimization techniques are required at the algorithm and hardware levels.

Tradeoffs between specialized processors and generalpurpose processors will continue to confound the industry for the foreseeable future. This may provide an opening for new technologies, memories, eFPGAs or other programmable logic or software, but there is still a long way for a solid ground in confident industry adoption. Security countermeasures should elevate as a first class constraint, moving from a subsequent concern in IC design, contrary to traditional goals involving cost, performance, and reliability [181]. Additionally, the tradeoff for opting for the best solution through a purpose-built processor for efficiency, or through an off-theshelf component will vary widely by application and ultimately by how these solutions perform over time and under load. Regardless, the inferencing market has opened the door to much different architectures and approaches than in the past, and there is no indication that will change anytime soon.

The rise of Internet of Things and edge systems and their use in large-scale, commercially sensitive applications makes attacks a growing concern for developers in all application domains. Many mitigation techniques come with major overheads in power performance and silicon die area that are impractical for IoT devices. On top, a growing concern involves how machine learning assists in securing IoT infrastructures, or if deep learning reverses the effects of countermeasures. Nevertheless, as machine learning based IDS obtained using hardware acceleration, compared to software, reaches high levels of accuracy, more that 95%, and boosts the classification speed significantly, open the way for integration at the IoT edge devices, which is especially challenging in real-time applications. Additionally, anomaly detection in IoT systems with transient behavior and, in domains that rapidly evolve becoming smarter (e.g., vehicles becoming more intelligent), is highly important and challenging, ultimately needing designs of effective and proactive secure IoT infrastructures. Moreover, it is also important to focus on developing IoT protection mechanisms that detect known and unknown attacks while being protocol-independent and non-cryptography related. Concentrating on the challenge of exploding unlabelled data in IoT and developing labelled IoT datasets for anomaly detection purposes, are also important research areas.

Hopefully, this article will be useful for academia and industry research, to identify the advantages and security drawbacks of different machine learning methods for an IoT infrastructure. Additionally, this survey will enable security and privacy designers enhance IoT devices countermeasures from traditional ones, while unleashing the development of efficient, low-latency, and reliable, ML-based intelligent services.

REFERENCES

- N. Chaabouni, M. Mosbah, A. Zemmari, C. Sauvignac, and P. Faruki, "Network intrusion detection for IoT security based on learning techniques," *IEEE Commun. Surveys Tuts.*, vol. 21, no. 3, pp. 2671–2701, 3rd Quart., 2019.
- [2] A. Humayed, J. Lin, F. Li, and B. Luo, "Cyber-physical systems security—A survey," *IEEE Internet Things J.*, vol. 4, no. 6, pp. 1802–1831, Dec. 2017.
- [3] D. Papp, Z. Ma, and L. Buttyan, "Embedded systems security: Threats, vulnerabilities, and attack taxonomy," in *Proc. 13th Annu. Conf. Privacy, Secur. Trust (PST)*, Jul. 2015, pp. 145–152.
- [4] K. Chen, S. Zhang, Z. Li, Y. Zhang, Q. Deng, S. Ray, and Y. Jin, "Internet-of-Things security and vulnerabilities: Taxonomy, challenges, and practice," *J. Hardw. Syst. Secur.*, vol. 2, no. 2, pp. 97–110, Jun. 2018, doi: 10.1007/s41635-017-0029-7.
- [5] J. Granjal, E. Monteiro, and J. S. Silva, "Security for the Internet of Things: A survey of existing protocols and open research issues," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 3, pp. 1294–1312, 3rd Quart., 2015.
- [6] ARM. (2018). Trusted Base System Architecture Client (4th Edition) System Hardware on ARM, Document Number: ARM Den 0021D. [Online]. Available: https://developer.arm.com/documentation/den0021/d/
- STM. (2021). STM32U575/585 ARM-Based 32-Bit MCUs, STM32U5 Series, RM0456. Accessed: Aug. 10, 2021. [Online]. Available: https://www.st.com/resource/en/reference_manual/rm0456stm32u575585-ar mbased-32bit-mcus-stmicroelectronics.pdf
- [8] K. Serebryany, "ARM memory tagging extension and how it improves C/C++ memory safety," in *Login USENIX Magazine*, vol. 44. Berkeley, CA, USA: USENIX Association, 2019.
- [9] P. Nasahl, R. Schilling, M. Werner, J. Hoogerbrugge, M. Medwed, and S. Mangard, "CrypTag: Thwarting physical and logical memory vulnerabilities using cryptographically colored memory," in *Proc. ACM Asia Conf. Comput. Commun. Secur.*, New York, NY, USA, May 2021, pp. 200–212, doi: 10.1145/3433210.3453684.
- [10] PCI Security Standards Council (PCI SSC). (2018). PCI DSS Quick Reference Guide Understanding the Payment Card Industry Data Security Standard Version 3.2.1. [Online]. Available: https://www. pcisecuritystandards.org/document_library
- [11] P. Fremantle, A Security Survey of Middleware for the Internet of Things (Security). London, U.K.: Institution of Engineering and Technology, 2016, pp. 1–31. [Online]. Available: https://digitallibrary.theiet.org/content/books/10.1049/pbse002e_ch1
- [12] I. Butun, A. Sari, and P. Österberg, "Hardware security of fog end-devices for the Internet of Things," *Sensors*, vol. 20, no. 20, p. 5729, Oct. 2020.
- [13] G. Kornaros, O. Tomoutzoglou, and M. Coppola, "Hardware-assisted security in electronic control units: Secure automotive communications by utilizing one-time-programmable network on chip and firewalls," *IEEE Micro*, vol. 38, no. 5, pp. 63–74, Sep. 2018.
- [14] D. Šišejković, F. Merchant, L. M. Reimann, R. Leupers, M. Giacometti, and S. Kegreiß, "A secure hardware-software solution based on RISC-V, logic locking and microkernel," in *Proc. 23th Int. Workshop Softw. Compilers Embedded Syst.*, 2020, pp. 62–65. [Online]. Available: https://doi.org/10.1145/3378678.3391886
- [15] G. Kornaros, E. Wozniak, O. Horst, N. Koch, C. Prehofer, A. Rigo, and M. Coppola, "Secure and trusted open CPS platforms," in *Solutions for Cyber-Physical Systems Ubiquity*, N. Druml, A. Genser, A. Hoeller, A. Krieg, and M. Menghin, Eds. Hershey, PA, USA: IGI Global, 2018, ch. 12, pp. 301–324, doi: 10.4018/978-1-5225-2845-6.ch012.
- [16] K. A. Stouffer, V. Y. Pillitteri, S. Lightman, M. Abrams, and A. Hahn, "Guide to industrial control systems (ICS) security," Special Publication (NIST SP), Nat. Inst. Standards Technol., Gaithersburg, MD, USA, Tech. Rep. NIST Special Publication 800-82, May 2015, doi: 10.6028/NIST.SP.800-82r2.
- [17] Y. Luo, Y. Xiao, L. Cheng, G. Peng, and D. Yao, "Deep learning-based anomaly detection in cyber-physical systems: Progress and opportunities," ACM Comput. Surv., vol. 54, no. 5, pp. 1–36, Jun. 2022, doi: 10.1145/3453155.
- [18] L. Xiao, X. Wan, X. Lu, Y. Zhang, and D. Wu, "IoT security techniques based on machine learning: How do IoT devices use AI to enhance security?" *IEEE Signal Process. Mag.*, vol. 35, no. 5, pp. 41–49, Sep. 2018.
- [19] M. Zolanvari, M. A. Teixeira, L. Gupta, K. M. Khan, and R. Jain, "Machine learning-based network vulnerability analysis of industrial Internet of Things," *IEEE Internet Things J.*, vol. 6, no. 4, pp. 6822–6834, Aug. 2019.

- [20] I. Andrea, C. Chrysostomou, and G. Hadjichristofi, "Internet of Things: Security vulnerabilities and challenges," in *Proc. IEEE Symp. Comput. Commun. (ISCC)*, Jul. 2015, pp. 180–187.
- [21] V. Hassija, V. Chamola, V. Saxena, D. Jain, P. Goyal, and B. Sikdar, "A survey on IoT security: Application areas, security threats, and solution architectures," *IEEE Access*, vol. 7, pp. 82721–82743, 2019.
- [22] R. Roman, J. Lopez, and M. Mambo, "Mobile edge computing, fog: A survey and analysis of security threats and challenges," *Future Gener. Comput. Syst.*, vol. 78, pp. 680–698, Jan. 2018. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0167739X16305635
- [23] A.-R. Sadeghi, C. Wachsmann, and M. Waidner, "Security and privacy challenges in industrial Internet of Things," in *Proc. 52nd Annu. Design Autom. Conf.*, Jun. 2015, pp. 1–6.
- [24] S. Bhattacharya and D. Mukhopadhyay, "Who watches the watchmen?: Utilizing performance monitors for compromising keys of RSA on Intel platforms," in *CHES*. New York, NY, USA: Association for Computing Machinery, 2015, pp. 248–266. [Online]. Available: https://www.iacr.org/archive/ches2015/92930241/92930241.pdf
- [25] E. Bout, V. Loscri, and A. Gallais, "How machine learning changes the nature of cyberattacks on IoT networks: A survey," *IEEE Commun. Surveys Tuts.*, vol. 24, no. 1, pp. 248–279, 1st Quart., 2022.
- [26] K. Eykholt, I. Evtimov, E. Fernandes, B. Li, A. Rahmati, C. Xiao, A. Prakash, T. Kohno, and D. Song, "Robust physical-world attacks on deep learning models," in *Proc. IEEE/CVF Conf. Comput. Vis. Pattern Recognit.*, 2018, pp. 1625–1634, doi: 10.1109/CVPR.2018.00175.
- [27] A. Kulow, T. Schamberger, L. Tebelmann, and G. Sigl, "Finding the needle in the haystack: Metrics for best trace selection in unsupervised sidechannel attacks on blinded RSA," *IEEE Trans. Inf. Forensics Security*, vol. 16, pp. 3254–3268, 2021.
- [28] K. Yang, Q. Li, X. Lin, X. Chen, and L. Sun, "IFinger: Intrusion detection in industrial control systems via register-based fingerprinting," *IEEE J. Sel. Areas Commun.*, vol. 38, no. 5, pp. 955–967, May 2020.
- [29] B. Deokar and A. Hazarnis, "Intrusion detection system using log files and reinforcement learning," *Int. J. Comput. Appl.*, vol. 45, no. 19, pp. 28–35, 2012.
- [30] T. T. Nguyen and V. J. Reddi, "Deep reinforcement learning for cyber security," *IEEE Trans. Neural Netw. Learn. Syst.*, early access, Nov. 13, 2021, doi: 10.1109/TNNLS.2021.3121870.
- [31] P. Freitas de Araujo-Filho, G. Kaddoum, D. R. Campelo, A. Gondim Santos, D. Macêdo, and C. Zanchettin, "Intrusion detection for cyber–physical systems using generative adversarial networks in fog environment," *IEEE Internet Things J.*, vol. 8, no. 8, pp. 6247–6256, Apr. 2021.
- [32] N. Patel, A. Sasan, and H. Homayoun, "Analyzing hardware based malware detectors," in *Proc. 54th Annu. Design Autom. Conf.*, Jun. 2017, pp. 1–6.
- [33] A. Tang, S. Sethumadhavan, and S. J. Stolfo, "Unsupervised anomalybased malware detection using hardware features," in *Research in Attacks, Intrusions and Defenses*, A. Stavrou, H. Bos, and G. Portokalidis, Eds. Cham, Switzerland: Springer, 2014, pp. 109–129.
- [34] X. Wang and R. Karri, "Reusing hardware performance counters to detect and identify kernel control-flow modifying rootkits," *IEEE Trans. Comput.-Aided Design Integr. Circuits Syst.*, vol. 35, no. 3, pp. 485–498, Aug. 2016.
- [35] Y. Liu, S. Garg, J. Nie, Y. Zhang, Z. Xiong, J. Kang, and M. S. Hossain, "Deep anomaly detection for time-series data in industrial IoT: A communication-efficient on-device federated learning approach," *IEEE Internet Things J.*, vol. 8, no. 8, pp. 6348–6358, Apr. 2021.
- [36] P. Maene, J. Götzfried, R. De Clercq, T. Müller, F. Freiling, and I. Verbauwhede, "Hardware-based trusted computing architectures for isolation and attestation," *IEEE Trans. Comput.*, vol. 67, no. 3, pp. 361–374, Mar. 2017.
- [37] M. Merenda, C. Porcaro, and D. Iero, "Edge machine learning for AIenabled IoT devices: A review," *Sensors*, vol. 20, no. 9, p. 2533, 2020.
- [38] Q. Xu, M. T. Arafin, and G. Qu, "Security of neural networks from hardware perspective: A survey and beyond," in *Proc. 26th Asia South Pacific Design Autom. Conf.*, Jan. 2021, pp. 449–454, doi: 10.1145/3394885.3431639.
- [39] R. Mitchell and I.-R. Chen, "A survey of intrusion detection techniques for cyber-physical systems," *ACM Comput. Surv.*, vol. 46, no. 4, pp. 1–29, Apr. 2014, doi: 10.1145/2542049.
- [40] M. A. Al-Garadi, A. Mohamed, A. K. Al-Ali, X. Du, I. Ali, and M. Guizani, "A survey of machine and deep learning methods for Internet of Things (IoT) security," *IEEE Commun. Surveys Tuts.*, vol. 22, no. 3, pp. 1646–1685, 3rd Quart., 2020.

- [41] Z. Huang, Q. Wang, Y. Chen, and X. Jiang, "A survey on machine learning against hardware trojan attacks: Recent advances and challenges," *IEEE Access*, vol. 8, pp. 10796–10826, 2020.
- [42] F. Hussain, R. Hussain, S. A. Hassan, and E. Hossain, "Machine learning in IoT security: Current solutions and future challenges," *IEEE Commun. Surveys Tuts.*, vol. 22, no. 3, pp. 1686–1721, 3rd Quart., 2020.
- [43] M. N. Khan, A. Rao, and S. Camtepe, "Lightweight cryptographic protocols for IoT-constrained devices: A survey," *IEEE Internet Things J.*, vol. 8, no. 6, pp. 4132–4156, Mar. 2021.
- [44] A. Khraisat and A. Alazab, "A critical review of intrusion detection systems in the Internet of Things: Techniques, deployment strategy, validation strategy, attacks, public datasets and challenges," *Cybersecurity*, vol. 4, no. 1, pp. 1–27, Dec. 2021.
- [45] S. Sidhu, B. J. Mohd, and T. Hayajneh, "Hardware security in IoT devices with emphasis on hardware trojans," *J. Sensor Actuator Netw.*, vol. 8, no. 3, p. 42, Aug. 2019.
- [46] A. Shawahna, S. M. Sait, and A. El-Maleh, "FPGA-based accelerators of deep learning networks for learning and classification: A review," *IEEE Access*, vol. 7, pp. 7823–7859, 2019.
- [47] K. Bochie, M. S. Gilbert, L. Gantert, M. S. M. Barbosa, D. S. V. Medeiros, and M. E. M. Campista, "A survey on deep learning for challenged networks: Applications and trends," *J. Netw. Comput. Appl.*, vol. 194, Nov. 2021, Art. no. 103213. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S1084804521002149
- [48] F. O. Olowononi, D. B. Rawat, and C. Liu, "Resilient machine learning for networked cyber physical systems: A survey for machine learning security to securing machine learning for CPS," *IEEE Commun. Surveys Tuts.*, vol. 23, no. 1, pp. 524–552, 1st Quart., 2021.
- [49] I. Rosenberg, A. Shabtai, Y. Elovici, and L. Rokach, "Adversarial machine learning attacks and defense methods in the cyber security domain," ACM Comput. Surv., vol. 54, no. 5, pp. 1–36, Jun. 2022, doi: 10.1145/3453158.
- [50] V. Chandola, A. Banerjee, and V. Kumar, "Anomaly detection: A survey," ACM Comput. Surv., vol. 41, no. 3, pp. 1–58, Jul. 2009, doi: 10.1145/1541880.1541882.
- [51] M. Suresh and R. Anitha, "Evaluating machine learning algorithms for detecting DDoS attacks," in *Advances in Network Security and Applications*, D. C. Wyld, M. Wozniak, N. Chaki, N. Meghanathan, and D. Nagamalai, Eds. Berlin, Germany: Springer, 2011, pp. 441–452.
- [52] M. Canizo, I. Triguero, A. Conde, and E. Onieva, "Multi-head CNN-RNN for multi-time series anomaly detection: An industrial case study," *Neurocomputing*, vol. 363, pp. 246–260, Oct. 2019. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0925231219309877
- [53] A. Theissler, "Detecting known and unknown faults in automotive systems using ensemble-based anomaly detection," *Knowl.-Based Syst.*, vol. 123, pp. 163–173, May 2017. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0950705117301077
- [54] G. Pachauri and S. Sharma, "Anomaly detection in medical wireless sensor networks using machine learning algorithms," *Proc. Comput. Sci.*, vol. 70, pp. 325–333, Dec. 2015. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S1877050915031907
- [55] R. Kumar, P. Kumar, and Y. Kumar, "Time series data prediction using IoT and machine learning technique," *Proc. Comput. Sci.*, vol. 167, pp. 373–381, Jan. 2020. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S1877050920307067
- [56] M. Munir, S. A. Siddiqui, A. Dengel, and S. Ahmed, "DeepAnT: A deep learning approach for unsupervised anomaly detection in time series," *IEEE Access*, vol. 7, pp. 1991–2005, 2019.
- [57] M. Yang, T. Zhu, B. Liu, Y. Xiang, and W. Zhou, "Machine learning differential privacy with multifunctional aggregation in a fog computing architecture," *IEEE Access*, vol. 6, pp. 17119–17129, 2018.
- [58] L. Fang, Y. Li, Z. Liu, C. Yin, M. Li, and Z. J. Cao, "A practical model based on anomaly detection for protecting medical IoT control services against external attacks," *IEEE Trans. Ind. Informat.*, vol. 17, no. 6, pp. 4260–4269, Jun. 2021.
- [59] Y. Cheng, Y. Xu, H. Zhong, and Y. Liu, "Leveraging semisupervised hierarchical stacking temporal convolutional network for anomaly detection in IoT communication," *IEEE Internet Things J.*, vol. 8, no. 1, pp. 144–155, Jan. 2021.
- [60] M. M. Hassan, S. Huda, S. Sharmeen, J. Abawajy, and G. Fortino, "An adaptive trust boundary protection for IIoT networks using deeplearning feature-extraction-based semisupervised model," *IEEE Trans. Ind. Informat.*, vol. 17, no. 4, pp. 2860–2870, Apr. 2021.

- [61] Y.-H. Chen, T.-J. Yang, J. Emer, and V. Sze, "Eyeriss V2: A flexible accelerator for emerging deep neural networks on mobile devices," *IEEE J. Emerg. Sel. Topics Circuits Syst.*, vol. 9, no. 2, pp. 292–308, Jun. 2019.
- [62] B. Moons, R. Uytterhoeven, W. Dehaene, and M. Verhelst, "14.5 envision: A 0.26-to-10TOPS/W subword-parallel dynamic-voltage-accuracyfrequency-scalable convolutional neural network processor in 28 nm FDSOI," in *IEEE Int. Solid-State Circuits Conf. (ISSCC) Dig. Tech. Papers*, Feb. 2017, pp. 246–247.
- [63] S. Lee and S. Nirjon, "Neuro.ZERO: A zero-energy neural network accelerator for embedded sensing and inference systems," in *Proc. 17th Conf. Embedded Netw. Sensor Syst.*, Nov. 2019, pp. 138–152, doi: 10.1145/3356250.3360030.
- [64] M. Horowitz, "1.1 Computing's energy problem (and what we can do about it)," in *IEEE Int. Solid-State Circuits Conf. (ISSCC) Dig. Tech. Papers*, Feb. 2014, pp. 10–14.
- [65] A. Gholami, S. Kim, Z. Dong, Z. Yao, M. W. Mahoney, and K. Keutzer, "A survey of quantization methods for efficient neural network inference," 2021, arXiv:2103.13630.
- [66] M. Drumond, T. Lin, M. Jaggi, and B. Falsafi, "Training DNNs with hybrid block floating point," in *Proc. 32nd Int. Conf. Neural Inf. Process. Syst.*, 2018, pp. 451–461.
- [67] J. Lee, C. Kim, S. Kang, D. Shin, S. Kim, and H.-J. Yoo, "UNPU: A 50.6 TOPS/W unified deep neural network accelerator with 1b-to-16b fully-variable weight bit-precision," in *IEEE Int. Solid-State Circuits Conf. (ISSCC) Dig. Tech. Papers*, Feb. 2018, pp. 218–220.
- [68] J. Gao, X. Song, Q. Wen, P. Wang, L. Sun, and H. Xu, "RobustTAD: Robust time series anomaly detection via decomposition and convolutional neural networks," 2020, arXiv:2002.09545.
- [69] F. N. Iandola, S. Han, M. W. Moskewicz, K. Ashraf, W. J. Dally, and K. Keutzer, "SqueezeNet: AlexNet-level accuracy with 50x fewer parameters and <0.5MB model size," 2016, arXiv:1602.07360.</p>
- [70] A. G. Howard, M. Zhu, B. Chen, D. Kalenichenko, W. Wang, T. Weyand, M. Andreetto, and H. Adam, "MobileNets: Efficient convolutional neural networks for mobile vision applications," 2017, arXiv:1704.04861.
- [71] T. Chen, Z. Du, N. Sun, J. Wang, C. Wu, Y. Chen, and O. Temam, "DianNao: A small-footprint high-throughput accelerator for ubiquitous machine-learning," in *Proc. 19th Int. Conf. Architectural Support Program. Lang. Operating Syst.*, Feb. 2014, pp. 269–284, doi: 10.1145/2541940.2541967.
- [72] D. Shin, J. Lee, J. Lee, J. Lee, and H.-J. Yoo, "DNPU: An energy-efficient deep-learning processor with heterogeneous multi-core architecture," *IEEE Micro*, vol. 38, no. 5, pp. 85–93, Sep. 2018.
- [73] B. Reagen, P. Whatmough, R. Adolf, S. Rama, H. Lee, S. K. Lee, J. M. Hernandez-Lobato, G.-Y. Wei, and D. Brooks, "Minerva: Enabling low-power, highly-accurate deep neural network accelerators," in *Proc. ACM/IEEE 43rd Annu. Int. Symp. Comput. Archit. (ISCA)*, Jun. 2016, pp. 267–278.
- [74] S. Cho, H. Choi, E. Park, H. Shin, and S. Yoo, "McDRAM V2: Indynamic random access memory systolic array accelerator to address the large model problem in deep neural networks on the edge," *IEEE Access*, vol. 8, pp. 135223–135243, 2020.
- [75] E. Viegas, A. O. Santin, A. França, R. Jasinski, V. A. Pedroni, and L. S. Oliveira, "Towards an energy-efficient anomaly-based intrusion detection engine for embedded systems," *IEEE Trans. Comput.*, vol. 66, no. 1, pp. 163–177, Jan. 2016.
- [76] X. Li and J. Lin, "Linear time complexity time series classification with bag-of-pattern-features," in *Proc. IEEE Int. Conf. Data Mining (ICDM)*, Nov. 2017, pp. 277–286.
- [77] H. Sayadi, Y. Gao, H. Mohammadi Makrani, J. Lin, P. C. Costa, S. Rafatirad, and H. Homayoun, "Towards accurate run-time hardwareassisted stealthy malware detection: A lightweight, yet effective time series CNN-based approach," *Cryptography*, vol. 5, no. 4, p. 28, Oct. 2021. [Online]. Available: https://www.mdpi.com/2410-387X/5/4/28
- [78] C. Moratelli, S. Johann, M. Neves, and F. Hessel, "Embedded virtualization for the design of secure IoT applications," in *Proc. Int. Symp. Rapid Syst. Prototyping (RSP)*, Oct. 2016, pp. 1–5.
- [79] N. S. Chockaiah, S. K. S. Kayal, J. K. Malar, P. Kirithika, and M. N. Devi, "Hardware trojan detection using machine learning technique," in *Proc. Int. Conf. Recent Trends Mach. Learn., IoT, Smart Cities Appl.*, V. K. Gunjan and J. M. Zurada, Eds. Singapore: Springer, 2021, pp. 415–423.

- [80] K. G. Liakos, G. K. Georgakilas, S. Moustakidis, P. Karlsson, and F. C. Plessas, "Machine learning for hardware trojan detection: A review," in *Proc. Panhellenic Conf. Electron. Telecommun. (PACET)*, Nov. 2019, pp. 1–6.
- [81] F. Yao, A. S. Rakin, and D. Fan, DeepHammer: Depleting the Intelligence of Deep Neural Networks Through Targeted Chain of Bit Flips. Berkeley, CA, USA: USENIX Association, 2020.
- [82] Y. Jin and Y. Makris, "Hardware trojan detection using path delay fingerprint," in *Proc. IEEE Int. Workshop Hardw.-Oriented Secur. Trust*, Jun. 2008, pp. 51–57.
- [83] Y. Hou, H. He, K. Shamsi, Y. Jin, D. Wu, and H. Wu, "On-chip analog trojan detection framework for microprocessor trustworthiness," *IEEE Trans. Comput.-Aided Design Integr. Circuits Syst.*, vol. 38, no. 10, pp. 1820–1830, Oct. 2019.
- [84] Y. Liu, K. Huang, and Y. Makris, "Hardware trojan detection through golden chip-free statistical side-channel fingerprinting," in *Proc. 51st Annu. Design Autom. Conf. Design Autom. Conf. (DAC)*, New York, NY, USA, 2014, pp. 1–6, doi: 10.1145/2593069.2593147.
- [85] C. Bao, D. Forte, and A. Srivastava, "On application of one-class SVM to reverse engineering-based hardware trojan detection," in *Proc. 15th Int. Symp. Quality Electron. Design*, Mar. 2014, pp. 47–54.
- [86] R. Sharma, V. S. Rathor, G. K. Sharma, and M. Pattanaik, "A new hardware trojan detection technique using deep convolutional neural network," *Integration*, vol. 79, pp. 1–11, Jul. 2021.
- [87] T. Iwase, Y. Nozaki, M. Yoshikawa, and T. Kumaki, "Detection technique for hardware trojans using machine learning in frequency domain," in *Proc. IEEE 4th Global Conf. Consum. Electron. (GCCE)*, Oct. 2015, pp. 185–186.
- [88] K. Hasegawa, M. Yanagisawa, and N. Togawa, "Trojan-feature extraction at gate-level netlists and its application to hardware-trojan detection using random forest classifier," in *Proc. IEEE Int. Symp. Circuits Syst. (ISCAS)*, May 2017, pp. 1–4.
- [89] F. K. Lodhi, I. Abbasi, F. Khalid, O. Hasan, F. Awwad, and S. R. Hasan, "A self-learning framework to detect the intruded integrated circuits," in *Proc. IEEE Int. Symp. Circuits Syst. (ISCAS)*, May 2016, pp. 1702–1705.
- [90] F. Khalid, S. R. Hasan, S. Zia, O. Hasan, F. Awwad, and M. Shafique, "MacLeR: Machine learning-based runtime hardware trojan detection in resource-constrained IoT edge devices," *IEEE Trans. Comput.-Aided Design Integr. Circuits Syst.*, vol. 39, no. 11, pp. 3748–3761, Nov. 2020, doi: 10.1109/TCAD.2020.3012236.
- [91] H. Zhao, L. Kwiat, K. A. Kwiat, C. A. Kamhoua, and L. Njilla, "Applying chaos theory for runtime hardware trojan monitoring and detection," *IEEE Trans. Dependable Secure Comput.*, vol. 17, no. 4, pp. 716–729, Jul. 2020.
- [92] X. Zhang and M. Tehranipoor, "Case study: Detecting hardware trojans in third-party digital IP cores," in *Proc. IEEE Int. Symp. Hardw.-Oriented Secur. Trust*, Jun. 2011, pp. 67–70.
- [93] A. Kulkarni, Y. Pino, M. French, and T. Mohsenin, "Real-time anomaly detection framework for many-core router through machine-learning techniques," ACM J. Emerg. Technol. Comput. Syst., vol. 13, no. 1, pp. 1–22, Dec. 2016, doi: 10.1145/2827699.
- [94] M. I. Mera Collantes, Z. Ghodsi, and S. Garg, "SafeTPU: A verifiably secure hardware accelerator for deep neural networks," in *Proc. IEEE* 38th VLSI Test Symp. (VTS), Apr. 2020, pp. 1–6.
- [95] J. Redmon, S. Divvala, R. Girshick, and A. Farhadi, "You only look once: Unified, real-time object detection," in *Proc. IEEE Conf. Comput. Vis. Pattern Recognit. (CVPR)*, 2016, pp. 779–788, doi: 10.1109/CVPR.2016.91.
- [96] Q.-C. Mao, H.-M. Sun, Y. B. Liu, and R.-S. Jia, "Mini-YOLOv3: Realtime object detector for embedded applications," *IEEE Access*, vol. 7, pp. 133529–133538, 2019.
- [97] M. Javaheripi, M. Samragh, G. Fields, T. Javidi, and F. Koushanfar, "CleaNN: Accelerated trojan shield for embedded neural networks," in *Proc. 39th Int. Conf. Comput.-Aided Design*, New York, NY, USA, Nov. 2020, doi: 10.1145/3400302.3415671.
- [98] F. Tramer and D. Boneh, "Slalom: Fast, verifiable and private execution of neural networks in trusted hardware," in *Proc. Int. Conf. Learn. Represent.*, 2019, pp. 1–19. [Online]. Available: https://openreview. net/forum?id=rJVorjCcKQ
- [99] A. Chakraborty, A. Mondai, and A. Srivastava, "Hardware-assisted intellectual property protection of deep learning models," in *Proc. 57th* ACM/IEEE Design Autom. Conf. (DAC), Jul. 2020, pp. 1–6.
- [100] H. Qiu, T. Dong, T. Zhang, J. Lu, G. Memmi, and M. Qiu, "Adversarial attacks against network intrusion detection in IoT systems," *IEEE Int. Things J.*, vol. 8, no. 13, pp. 10327–10335, Jul. 2021.

- [101] H. Guo, L. Peng, J. Zhang, F. Qi, and L. Duan, "Fooling AI with AI: An accelerator for adversarial attacks on deep learning visual classification," in *Proc. IEEE 30th Int. Conf. Appl.-Specific Syst., Archit. Processors (ASAP)*, Los Alamitos, CA, USA, Jul. 2019, p. 136, doi: 10.1109/asap.2019.00-16.
- [102] A. Vashist, A. Keats, S. M. Pudukotai Dinakarrao, and A. Ganguly, "Securing a wireless network-on-chip against jamming-based denial-ofservice and eavesdropping attacks," *IEEE Trans. Very Large Scale Integr.* (VLSI) Syst., vol. 27, no. 12, pp. 2781–2791, Dec. 2019.
- [103] K. Z. Snow, F. Monrose, L. Davi, A. Dmitrienko, C. Liebchen, and A. Sadeghi, "Just-in-time code reuse: On the effectiveness of fine-grained address space layout randomization," in *Proc. IEEE Symp. Secur. Pri*vacy, May 2013, pp. 574–588.
- [104] L. Zhou, Y. Hu, and Y. Makris, "A hardware-based architecture-neutral framework for real-time iot workload forensics," *IEEE Trans. Comput.*, vol. 69, no. 11, pp. 1668–1680, Nov. 2020.
- [105] P. Krishnamurthy, R. Karri, and F. Khorrami, "Anomaly detection in realtime multi-threaded processes using hardware performance counters," *IEEE Trans. Inf. Forensics Security*, vol. 15, pp. 666–680, 2020.
- [106] Y. Han, S. Etigowni, H. Liu, S. Zonouz, and A. Petropulu, "Watch me, but Don't touch me! Contactless control flow monitoring via electromagnetic emanations," in *Proc. ACM SIGSAC Conf. Comput. Commun. Secur.*, Oct. 2017, pp. 1095–1108, doi: 10.1145/3133956.3134081.
- [107] O. Meynard, D. Réal, S. Guilley, F. Flament, J.-L. Danger, and F. Valette, "Characterization of the electromagnetic side channel in frequency domain," in *Information Security and Cryptology*, X. Lai, M. Yung, and D. Lin, Eds. Berlin, Germany: Springer, 2011, pp. 471–486.
- [108] C. Whitnall and E. Oswald, "A critical analysis of ISO 17825 ('testing methods for the mitigation of non-invasive attack classes against cryptographic modules')," in *Advances in Cryptology–(ASIACRYP)*, S. D. Galbraith and S. Moriai, Eds. Cham, Switzerland: Springer, 2019, pp. 256–284.
- [109] P. C. Kocher, "Timing attacks on implementations of Diffie-Hellman, RSA, DSS, and other systems," in *Advances in Cryptology–(CRYPTO)*, N. Koblitz, Ed. Berlin, Germany: Springer, 1996, pp. 104–113.
- [110] H. Haddadi, V. Christophides, R. Teixeira, K. Cho, S. Suzuki, and A. Perrig, "SIOTOME: An edge-ISP collaborative architecture for IoT security," in *Proc. 1st Int. Workshop Secur. Privacy Internet Things* (*IoTSec*), 2018, pp. 42–45.
- [111] I. Martinez, A. S. Hafid, and A. Jarray, "Design, resource management, and evaluation of fog computing systems: A survey," *IEEE Internet Things J.*, vol. 8, no. 4, pp. 2494–2516, Feb. 2021.
- [112] B. Donassolo, I. Fajjari, A. Legrand, and P. Mertikopoulos, "Fog based framework for IoT service provisioning," in *Proc. 16th IEEE Annu. Consum. Commun. Netw. Conf. (CCNC)*, Jan. 2019, pp. 1–6.
- [113] I. Azimi, A. Anzanpour, A. M. Rahmani, T. Pahikkala, M. Levorato, P. Liljeberg, and N. Dutt, "HiCH: Hierarchical fog-assisted computing architecture for healthcare IoT," ACM Trans. Embedded Comput. Syst., vol. 16, no. 5s, pp. 1–20, Oct. 2017, doi: 10.1145/3126501.
- [114] J. Canedo and A. Skjellum, "Using machine learning to secure IoT systems," in *Proc. 14th Annu. Conf. Privacy, Secur. Trust (PST)*, Dec. 2016, pp. 219–222.
- [115] H. HaddadPajouh, A. Dehghantanha, R. Khayami, and K.-K. R. Choo, "A deep Recurrent Neural Network based approach for Internet of Things malware threat hunting," *Future Gener. Comput. Syst.*, vol. 85, pp. 88–96, Aug. 2018. [Online]. Available: https://www. sciencedirect.com/science/article/pii/S0167739X1732486X
- [116] S. Deng, Z. Xiang, P. Zhao, J. Taheri, H. Gao, J. Yin, and A. Y. Zomaya, "Dynamical resource allocation in edge for trustable Internet-of-Things systems: A reinforcement learning method," *IEEE Trans. Ind. Informat.*, vol. 16, no. 9, pp. 6103–6113, Sep. 2020.
- [117] A. Azmoodeh, A. Dehghantanha, and K.-K. R. Choo, "Robust malware detection for internet of (battlefield) things devices using deep Eigenspace learning," *IEEE Trans. Sustain. Comput.*, vol. 4, no. 1, pp. 88–95, Jan./Mar. 2019.
- [118] A. Azmoodeh, A. Dehghantanha, M. Conti, and K.-K. R. Choo, "Detecting crypto-ransomware in IoT networks based on energy consumption footprint," *J. Ambient Intell. Humanized Comput.*, vol. 9, no. 4, pp. 1141–1152, 2017.
- [119] Z. Pan, J. Sheldon, C. Sudusinghe, S. Charles, and P. Mishra, "Hardwareassisted malware detection using machine learning," in *Proc. Design*, *Autom. Test Eur. Conf. Exhib. (DATE)*, Feb. 2021, pp. 1775–1780.
- [120] L. Zhou, Y. Zhang, and Y. Makris, "TPE: A hardware-based TLB profiling expert for workload reconstruction," *IEEE J. Emerg. Sel. Topics Circuits Syst.*, vol. 11, no. 2, pp. 292–305, Jun. 2021.

- [121] S. P. Kadiyala, P. Jadhav, S.-K. Lam, and T. Srikanthan, "Hardware performance counter-based fine-grained malware detection," *ACM Trans. Embedded Comput. Syst.*, vol. 19, no. 5, pp. 1–17, Sep. 2020, doi: 10.1145/3403943.
- [122] H. Wang, H. Sayadi, G. Kolhe, A. Sasan, S. Rafatirad, and H. Homayoun, "Phased-guard: Multi-phase machine learning framework for detection and identification of zero-day microarchitectural side-channel attacks," in *Proc. IEEE 38th Int. Conf. Comput. Design (ICCD)*, Oct. 2020, pp. 648–655.
- [123] B. Zhou, A. Gupta, R. Jahanshahi, M. Egele, and A. Joshi, "Hardware performance counters can detect malware: Myth or fact?" in *Proc. Asia Conf. Comput. Commun. Secur.*, May 2018, pp. 457–468, doi: 10.1145/3196494.3196515.
- [124] M. Ozsoy, C. Donovick, I. Gorelik, N. Abu-Ghazaleh, and D. Ponomarev, "Malware-aware processors: A framework for efficient online malware detection," in *Proc. IEEE 21st Int. Symp. High Perform. Comput. Archit.* (HPCA), Feb. 2015, pp. 651–661.
- [125] H. Sayadi, N. Patel, M. P. D. Sai, A. Sasan, S. Rafatirad, and H. Homayoun, "Ensemble learning for effective run-time hardware-based malware detection: A comprehensive analysis and classification," in *Proc. 55th* ACM/ESDA/IEEE Design Autom. Conf. (DAC), Jun. 2018, pp. 1–6.
- [126] A. B. Chowdhury, A. Mahapatra, D. Soni, and R. Karri, "Fuzzing+Hardware performance counters-based detection of algorithm subversion attacks on post-quantum signature schemes," *IEEE Trans. Comput.-Aided Design Integr. Circuits Syst.*, early access, Mar. 15, 2022, doi: 10.1109/TCAD.2022.3159749.
- [127] A. L. P. D. Franca, R. P. Jasinski, V. A. Pedroni, and A. O. Santin, "Moving network protection from software to hardware: An energy efficiency analysis," in *Proc. IEEE Comput. Soc. Annu. Symp. (VLSI)*, Jul. 2014, pp. 456–461.
- [128] T. Soylu, O. Erdem, A. Carus, and E. S. Guner, "Simple CART based real-time traffic classification engine on FPGAs," in *Proc. Int. Conf. Reconfigurable Comput. FPGAs (ReConFig)*, Dec. 2017, pp. 1–8.
- [129] G.-I. Trouli and G. Kornaros, "Automotive virtual in-sensor analytics for securing vehicular communication," *IEEE Design Test*, vol. 37, no. 3, pp. 91–98, Jun. 2020.
- [130] M. Elnawawy, A. Sagahyroon, and T. Shanableh, "FPGA-based network traffic classification using machine learning," *IEEE Access*, vol. 8, pp. 175637–175650, 2020.
- [131] X. Chang, H. Pan, W. Lin, and H. Gao, "A mixed-pruning based framework for embedded convolutional neural network acceleration," *IEEE Trans. Circuits Syst. I, Reg. Papers*, vol. 68, no. 4, pp. 1706–1715, Apr. 2021.
- [132] A. Das, D. Nguyen, J. Zambreno, G. Memik, and A. Choudhary, "An FPGA-based network intrusion detection architecture," *IEEE Trans. Inf. Forensics Security*, vol. 3, no. 1, pp. 118–132, Mar. 2008.
- [133] BrainChip. (2021). Akida Neural Processor SOC. Accessed: Aug. 20, 2021. [Online]. Available: https://brainchipinc. com/akida-neural-processor-soc/
- [134] Nepes AI. NM500 User'Ts Manual Version 1.6.3, by General Vision and Nepes Korea. Accessed: Sep. 1, 2021. [Online]. Available: http://www.theneuromorphic.com/nm500/
- [135] C. I. Nwakanma, J.-W. Kim, J.-M. Lee, and D.-S. Kim, "Edge AI prospect using the NeuroEdge computing system: Introducing a novel neuromorphic technology," *ICT Exp.*, vol. 7, no. 2, pp. 152–157, Jun. 2021. [Online]. Available: https://www.sciencedirect. com/science/article/pii/S240595952100059X
- [136] ARM. (2020). Latest NPU Adds to Arm's AI Platform Performance, Applicability, and Efficiency. Accessed: Aug. 1, 2021. [Online]. Available: https://www.arm.com/company/news/2020/10/latest-npu-adds-toarm-ai-platform-performance
- [137] NVIDIA. Nvidia Drive Hardware. Accessed: Aug. 1, 2021. [Online]. Available: https://www.nvidia.com/en-us/self-driving-cars/ drive-platform/hardware/
- [138] Renesas. (2022). e-AI Solution. Accessed: Aug. 1, 2021. [Online]. Available: https://www.renesas.com/us/en/application/keytechnology/artificial-intelligence/e-ai
- [139] K. Matsubara, T. Nagasawa, Y. Kaneda, H. Mitani, H. Sato, T. Iwase, Y. Aoki, K. Maekawa, H. Yamakoshi, T. Ito, H. Kondo, and T. Kono, "A 65 nm silicon-on-thin-box (SOTB) embedded 2T-MONOS flash achieving 0.22 pJ/bit read energy with 64 MHz access for IoT applications," in *Proc. Symp. VLSI Circuits*, Jun. 2019, pp. C202–C203.
- [140] ARM. (2020). Arm is Powering Innovation Through Artificial Intelligence. Accessed: Aug. 1, 2021. [Online]. Available: https://www. arm.com/solutions/artificial-intelligence

- [141] STM. (2021). X-CUBE-AI—AI AI Expansion Pack for STM32CubeMX. Accessed: Aug. 1, 2021. [Online]. Available: https://www.st.com/en/embedded-software/x-cube-ai.html#overview
- [142] L. Lai, N. Suda, and V. Chandra, "CMSIS-NN: Efficient neural network kernels for arm cortex-M CPUs," 2018, arXiv:1801.06601.
- [143] STM. (2021). FP-AI-NANOEDG1, Artificial Intelligence (AI) condition monitoring function pack for STM32Cube. Accessed: Aug. 30, 2021. [Online]. Available: https://www.st.com/en/embedded-software/fp-ainanoedg1.html
- [144] STM Life.Augmented. (Feb. 2022). LSM6DSOX: Machine Learning Core. Accessed: Mar. 10, 2022. [Online]. Available: https://www.st. com/en/mems-and-sensors/lsm6dsox.html
- [145] Greenwaves Technologies. Ultra low power GAP Processors. Accessed: Sep. 1, 2021. [Online]. Available: https://greenwaves-technologies. com/low-power-processor/
- [146] A. Burrello, M. Scherer, M. Zanghieri, F. Conti, and L. Benini, "A microcontroller is all you need: Enabling transformer execution on low-power IoT endnodes," in *Proc. IEEE Int. Conf. Omni-Layer Intell. Syst. (COINS)*, Aug. 2021, pp. 1–6.
- [147] A. Burrello, A. Garofalo, N. Bruschi, G. Tagliavini, D. Rossi, and F. Conti, "DORY: Automatic end-to-end deployment of real-world DNNs on low-cost IoT MCUs," *IEEE Trans. Comput.*, vol. 70, no. 8, pp. 1253–1268, Aug. 2021.
- [148] S. Liu, Z. Du, J. Tao, D. Han, T. Luo, Y. Xie, Y. Chen, and T. Chen, "Cambricon: An instruction set architecture for neural networks," in *Proc.* ACM/IEEE 43rd Annu. Int. Symp. Comput. Archit. (ISCA), Jun. 2016, pp. 393–405.
- [149] M. Rusci, M. Fariselli, A. Capotondi, and L. Benini, Leveraging AutomatedMixed-Low-Precision Quantization for Tiny Edge Microcontrollers. Cham, Switzerland: Springer, 2020.
- [150] P.-E. Novac, G. Boukli Hacene, A. Pegatoquet, B. Miramond, and V. Gripon, "Quantization and deployment of deep neural networks on microcontrollers," *Sensors*, vol. 21, no. 9, p. 2984, Apr. 2021.
- [151] D. Xu, C. Liu, Y. Wang, K. Tu, B. He, and L. Zhang, "Accelerating generative neural networks on unmodified deep learning processors—A software approach," *IEEE Trans. Comput.*, vol. 69, no. 8, pp. 1172–1184, Aug. 2020.
- [152] J. Woodruff, R. N. M. Watson, D. Chisnall, S. W. Moore, J. Anderson, B. Davis, B. Laurie, P. G. Neumann, R. Norton, and M. Roe, "The CHERI capability model: Revisiting RISC in an age of risk," in *Proc. ACM/IEEE 41st Int. Symp. Comput. Archit. (ISCA)*, Jun. 2014, pp. 457–468.
- [153] Qualcomm Technologies. (2017). Pointer Authentication on ARMv8.3. [Online]. Available: https://www.qualcomm.com/media/documents/files/ whitepaper-pointer-authentication-on-armv8-3.pdf
- [154] T. Nyman, G. Dessouky, S. Zeitouni, A. Lehikoinen, A. Paverd, N. Asokan, and A.-R. Sadeghi, "HardScope: Hardening embedded systems against data-oriented attacks," in *Proc. 56th ACM/IEEE Design Automat. Conf. (DAC)*, Jun. 2019, pp. 1–6.
- [155] S. P. Bayerl, T. Frassetto, P. Jauernig, K. Riedhammer, A.-R. Sadeghi, T. Schneider, E. Stapf, and C. Weinert, "Offline model guard: Secure and private ML on mobile devices," in *Proc. Design, Autom. Test Eur. Conf. Exhib. (DATE)*, Mar. 2020, pp. 460–465.
- [156] F. Brasser, D. Gens, P. Jauernig, A.-R. Sadeghi, and E. Stapf, "SANC-TUARY: ARMing TrustZone with user-space enclaves," in *Proc. Netw. Distrib. Syst. Secur. Symp.*, 2019, pp. 1–15.
- [157] R. Liu, L. Garcia, Z. Liu, B. Ou, and M. Srivastava, "SecDeep: Secure and performant on-device deep learning inference framework for mobile and IoT devices," in *Proc. Int. Conf. Internet–Things Design Implement.*, 2021, pp. 67–79.
- [158] Z. Sun, R. Sun, C. Liu, A. Roy Chowdhury, S. Jha, and L. Lu, "ShadowNet: A secure and efficient system for on-device model inference," 2020, arXiv:2011.05905.
- [159] F. Mo, A. S. Shamsabadi, K. Katevas, S. Demetriou, I. Leontiadis, A. Cavallaro, and H. Haddadi, "DarkneTZ: Towards model privacy at the edge using trusted execution environments," in *Proc. 18th Int. Conf. Mobile Syst., Appl., Services*, 2020, pp. 161–174.
- [160] J. Amacher and V. Schiavoni, "On the performance of arm trustzone," in *Proc. IFIP Int. Conf. Distrib. Appl. Interoperable Syst. (DAIS)*, 2019, pp. 133–151.
- [161] S. Zhao, Q. Zhang, Y. Qin, W. Feng, and D. Feng, "Minimal kernel: An operating system architecture for TEE to resist board level physical attacks," in *Proc. 22nd Int. Symp. Res. Attacks, Intrusions Defenses (RAID)*, Sep. 2019, pp. 105–120. [Online]. Available: https://www.usenix.org/conference/raid2019/presentation/zhao

- [162] D. Hwang, S. Yeleuov, J. Seo, M. Chung, H. Moon, and Y. Paek, "Ambassy: A runtime framework to delegate trusted applications in an ARM/FPGA hybrid system," *IEEE Trans. Mobile Comput.*, early access, Jun. 3, 2021, doi: 10.1109/TMC.2021.3086143.
- [163] O. Mutlu. (2021). Intelligent architectures for Intelligent Machines. [Online]. Available: https://youtu.be/nloY_jTJtU8?list= PL5Q2soXY2Zi8D_5MGV6EnXEJHnV2YFBJI
- [164] O. Mutlu, S. Ghose, J. Gómez-Luna, and R. Ausavarungnirun, "A modern primer on processing in memory," 2020, arXiv:2012.03112.
- [165] S. Pontarelli, G. Bianchi, and S. Teofili, "Traffic-aware design of a highspeed FPGA network intrusion detection system," *IEEE Trans. Comput.*, vol. 62, no. 11, pp. 2322–2334, Nov. 2013.
- [166] S. M. Tahsien, H. Karimipour, and P. Spachos, "Machine learning based solutions for security of Internet of Things (IoT): A survey," J. Netw. Comput. Appl., vol. 161, Jul. 2020, Art. no. 102630. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S1084804520301041
- [167] K. Tange, M. De Donno, X. Fafoutis, and N. Dragoni, "A systematic survey of industrial Internet of Things security: Requirements and fog computing opportunities," *IEEE Commun. Surveys Tuts.*, vol. 22, no. 4, pp. 2489–2520, 4th Quart., 2020.
- [168] A. Diro and N. Chilamkurti, "Leveraging LSTM networks for attack detection in fog-to-things communications," *IEEE Commun. Mag.*, vol. 56, no. 9, pp. 124–130, Sep. 2018.
- [169] A. Samy, H. Yu, and H. Zhang, "Fog-based attack detection framework for Internet of Things using deep learning," *IEEE Access*, vol. 8, pp. 74571–74585, 2020.
- [170] S. Das, Y. Liu, W. Zhang, and M. Chandramohan, "Semanticsbased online malware detection: Towards efficient real-time protection against malware," *IEEE Trans. Inf. Forensics Security*, vol. 11, no. 2, pp. 289–302, Feb. 2016.
- [171] G. Kornaros and D. Pnevmatikatos, "A survey and taxonomy of onchip monitoring of multicore systems-on-chip," ACM Trans. Design Autom. Electron. Syst., vol. 18, no. 2, pp. 1–38, Mar. 2013, doi: 10.1145/2442087.2442088.
- [172] A. Sengupta, S. Bhadauria, and S. P. Mohanty, "TL-HLS: Methodology for low cost hardware trojan security aware scheduling with optimal loop unrolling factor during high level synthesis," *IEEE Trans. Comput.-Aided Design Integr. Circuits Syst.*, vol. 36, no. 4, pp. 655–668, Apr. 2017.
- [173] S. I. Venieris, A. Kouris, and C.-S. Bouganis, "Toolflows for mapping convolutional neural networks on FPGAs: A survey and future directions," ACM Comput. Surv., vol. 51, no. 3, pp. 1–39, May 2019, doi: 10.1145/3186332.
- [174] S. Lahti, P. Sjövall, J. Vanne, and T. D. Hämäläinen, "Are we there yet? A study on the state of high-level synthesis," *IEEE Trans. Comput.-Aided Design Integr. Circuits Syst.*, vol. 38, no. 5, pp. 898–911, May 2019.

- [175] P. Mantovani, D. Giri, G. Di Guglielmo, L. Piccolboni, J. Zuckerman, E. G. Cota, M. Petracca, C. Pilato, and L. P. Carloni, "Agile SoC development with open ESP," in *Proc. 39th Int. Conf. Computer-Aided Design*, Nov. 2020, pp. 1–9.
- [176] F. Fahim, B. Hawks, C. Herwig, and J. Hirschauer, "Hls4ml: An opensource codesign workflow to empower scientific low-power machine learning devices," 2021, arXiv:2103.05579.
- [177] V. Duddu, "A survey of adversarial machine learning in cyber warfare," *Defence Sci. J.*, vol. 68, no. 4, pp. 356–366, 2018. [Online]. Available: https://publications.drdo.gov.in/ojs/index.php/dsj/article/view/12371
- [178] T. Chen, J. Liu, Y. Xiang, W. Niu, E. Tong, and Z. Han, "Adversarial attack and defense in reinforcement learning-from AI security view," *Cybersecurity*, vol. 2, no. 1, Dec. 2019.
- [179] N. Carlini and D. Wagner, "Adversarial examples are not easily detected: Bypassing ten detection methods," in *Proc. 10th ACM Workshop Artif. Intell. Secur.* New York, NY, USA: Association for Computing Machinery, 2017, pp. 3–14, doi: 10.1145/3128572.3140444.
- [180] W. Guo, D. Mu, J. Xu, P. Su, G. Wang, and X. Xing, "LEMNA: Explaining deep learning based security applications," in *Proc. ACM SIGSAC Conf. Comput. Commun. Secur.*, Oct. 2018, pp. 364–379, doi: 10.1145/3243734.3243792.
- [181] M. Rostami, F. Koushanfar, and R. Karri, "A primer on hardware security: Models, methods, and metrics," *Proc. IEEE*, vol. 102, no. 8, pp. 1283–1295, Aug. 2014.



GEORGIOS KORNAROS is an Associate Professor with the Electrical and Computer Engineering Department, Hellenic Mediterranean University, Greece, where he leads the Intelligent Systems and Computer Architecture Group. He was a System Architect of single-chip network processors for the industry and currently he is involved in multiple European research projects. He has published more than 60 scientific articles, and edited the book *MultiCore Embedded Systems* and holds

three patents. His research interests include multi-/many-core systems, security, high-speed communication architectures, and energy-efficient and heterogeneous computing. He is a member of the Technical Chamber of Greece.